

University of Texas at Tyler Scholar Works at UT Tyler

Mechanical Engineering Theses

Mechanical Engineering

Winter 12-11-2023

# RESIDENTIAL INDOOR AIR QUALITY STUDY – ASSESSMENT AND MITIGATION

Giovanni Cerrato

Follow this and additional works at: https://scholarworks.uttyler.edu/me\_grad

Part of the Mechanical Engineering Commons

#### **Recommended Citation**

Cerrato, Giovanni, "RESIDENTIAL INDOOR AIR QUALITY STUDY – ASSESSMENT AND MITIGATION" (2023). *Mechanical Engineering Theses.* Paper 28. http://hdl.handle.net/10950/4477

This Thesis is brought to you for free and open access by the Mechanical Engineering at Scholar Works at UT Tyler. It has been accepted for inclusion in Mechanical Engineering Theses by an authorized administrator of Scholar Works at UT Tyler. For more information, please contact tgullings@uttyler.edu.



# RESIDENTIAL INDOOR AIR QUALITY STUDY – ASSESSMENT AND MITIGATION

by

#### GIOVANNI JOSUE CERRATO

A thesis submitted in partial fulfillment of the requirements for the degree of Master of Science in Mechanical Engineering Department of Mechanical Engineering

Nelson Fumo, Ph.D., Committee Chair

College of Engineering

The University of Texas at Tyler September 2023 The University of Texas at Tyler Tyler, Texas

This is to certify that the Master's Thesis of

### GIOVANNI JOSUE CERRATO

has been approved for the thesis/dissertation requirement on (07 September 2023) for the Master of Science in Mechanical Engineering degree

Approvals:

DocuSigned by: Alson Funo -9BAB5E76B5E14AE.

Thesis Chair: Nelson Fumo, Ph.D.

DocuSigned by: Shik-Feng Chou

Member: Shih-Feng Chou, Ph.D.

-DocuSigned by: Nael Barakat

Member: Nael Barakat, Ph.D. and Chair, Department of Mechanical Engineering

DocuSigned by: very Mallon -FD302E384F154FA

Dean, College of Engineering

#### ACKNOWLEDGEMENTS

I would like to thank Dr. Nelson Fumo for serving as my thesis chair and advisor in my graduate studies. Specifically, for his patience in developing a professional character in me, encouragement, an abundance of technical guidance, and pushing me to new limits. He is a dedicated professional who has provided me with valuable lessons to take for the rest of my career. I would also like to thank Dr. Nael Barakat and Dr. Shih-Feng Chou for serving as members of my thesis committee. I appreciate their time and guidance in finalizing my thesis. Finally, I would like to thank every professional I had the opportunity to work with at the Trane Residential Heating and Cooling Research Lab. Their experience, feedback, and contributions were key in completing the research performed for this thesis.

Abstract		8
Preface		9
Chapter 1 – Intr	roduction	10
1.1 Overview	of Indoor Air Quality	10
1.2 Overview	v of ozone (O <sub>3</sub> )	10
1.3 Overview	v of VOCs	12
1.4 Overview	of IAQ mitigation	13
1.5 Overview	of electro-static and photocatalytic oxidizing air purifiers	13
1.6 Thesis out	tline and purpose	14
Chapter 2 – Lite	erature Review	17
2.1 Literature	e of ozone and ES air cleaners	17
2.1.1 Chara	acterization of the tropospheric boundary layer and outdoor ozone	17
2.1.2 Diurn	nal and seasonal characteristics of outdoor ozone	
2.1.3 Infiltr	ration of outdoor ozone into an indoor environment	19
2.1.4 Gener	ration of ozone from ES air cleaners	
2.2 Literature	e of VOCs and PCO air cleaner testing	
2.2.1 Overv	view of experimental procedures used in the testing of PCO air cleaners	
2.2.2 Labor	ratory testing of PCO air cleaners	
2.2.3 Field	testing of PCO air cleaners	

# TABLE OF CONTENTS

Chapter 3 – Methodology and results for the evaluation of ozone production from tested electro-
static air cleaners
3.1 Testing facilities for ozone experiments
3.2 Ozone monitoring
3.3 Overview of ozone experiments
3.4 Results and discussion of ozone experiments 40
3.4.1 Baseline indoor ozone levels in UTT house #2 (Ozone experiment 1) 40
3.4.2 Comparison of baseline indoor ozone levels in the UTT test houses (Ozone experiment
2)
3.4.3 Indoor ozone levels with mechanical ventilation in UTT house #2 (Ozone experiment
3)
3.4.4 Indoor ozone levels with mechanical ventilation and the powering on of two ES air
cleaners in UTT test house #2 (Ozone experiment 4) 44
3.4.5 Indoor ozone levels with the powering on of ES air cleaners in UTT test house #2 (Ozone
experiment 5 and 6)
3.5 Conclusions from ozone experiments 50
Chapter 4 – Correction model for climatic variables affecting voc readings
4.1 Overview of multi-pixel gas sensors
4.2 Effect of temperature and humidity on VOC sensor readings
4.3 Methodology for VOC correction model
4.4 Comparison of corrected data and uncorrected data

4.6 Conclusion of climatic correction model	61
Chapter 5 – Methodology and results for the evaluation of PCO effectivness	63
5.1 Testing facilities for VOC experiments	63
5.2 Overview of AWAIR IAQ sensors	63
5.3 Experimental methods for VOC injection	65
5.4 Experimental conditions of the research houses during testing	71
5.5 Specifications of tested PCO air cleaners	72
5.6 Evaluation of PCO air cleaner methodology	73
5.7 Results and discussion of VOC experiments	73
5.7.1 Comparison of natural decays to decays under the operation of a PCO	73
5.7.2 Evaluation of PCO performance	80
5.8 Conclusions from VOC experiments	83
Chapter 6 – Conclusions and recommendations	86
References	88
Appendix A – Ozone monitor zero check	99
Appendix B – Justification for the horizontal and vertical placement of the ozone monitor	100
Appendix C – AWAIR sensor recalibration process	101
Appendix D – Justification for horizontal and vertical placement of AWAIR IAQ sensors	102
Appendix E – Verification plots for climatic correction model	103

# LIST OF TABLES

Table 1. Maximum and minimum VOC concentrations in a diurnal cycle for four seasons [26] 1	18
Table 2. Construction of envelopes 1, 2, 3, and 4 [28]    2	20
Table 3. Characteristics of the bench, pilot, and full-scale setups [41]	26
Table 4. Removal efficiencies for bench & pilot scale setup [36]	27
Table 5. Removal efficiencies for bench & pilot scale setup [41]	29
Table 6. Removal efficiencies for bench & full-scale setup [41]    2	29
Table 7. Variations in VOC concentration for the powering on of the E8 PCO [42]	33
Table 8. Variations in VOC concentration for the powering on of the E15 PCO [42]	33
Table 9. A summary of the setup for all six ozone experiments	39
Table 10. Overview of correction model procedure    5	57
Table 11. Example of correction model procedure	58
Table 12. General procedure of experiments for PCO on/off status	66
Table 13. Description of VOC experiments with duration    7	71
Table 14. Specifications of PCO A and B    7	72
Table 15. Initial temperature and humidity values for all VOC experiments	74
Table 16. Initial VOC concentrations for all VOC experiments    7	75
Table 17. CADR and Single-Pass Removal Efficiency of PCO A and B       8	81
Table 18. Critical Point of ozone Test    10	00
Table 19. Stratification of ozone test    10	00

## LIST OF FIGURES

Figure 1. Test houses at the University of Texas at Tyler
Figure 2. Stool setup for the ozone monitor
Figure 3. Ladder setup for the ozone monitor
Figure 4. Ozone monitor location and nearest supply registers in house layout
Figure 5. Ozone experiment 1 results for test house #2
Figure 6. Ozone experiment 2 results for test houses #1 and #2
Figure 7. Normalized baseline indoor ozone levels for test house #1 (0-48 hours) and house #2
(48-168 hours)
Figure 8. Ozone experiment 3 results during ES air cleaner A's testing
Figure 9. Ozone experiment 3 results during ES air cleaner B's testing
Figure 10. Ozone experiment 4 results during ES air cleaner A's testing
Figure 11. Ozone experiment 4 results during ES air cleaner B's testing
Figure 12. Ozone experiment 5 results during ES air cleaner A's testing
Figure 13. Ozone experiment 5 results during ES air cleaner B's testing
Figure 14. Ozone experiment 6 results during ES air cleaner A's testing
Figure 15. Ozone experiment 6 results during ES air cleaner B's testing
Figure 16. Variations in VOC concentrations due to temperature and humidity
Figure 17. Uncorrected VOC data 55
Figure 18. Calculation of slope to be used in correction model procedure
Figure 19. Corrected VOC data 58
Figure 20. Curve fit of corrected data including verification data
Figure 21. Uncorrected sensor VOC readings versus corrected data

Figure 22. Stool setup for the AWAIR IAQ sensors
Figure 23. AWAIR sensor location and nearest supply registers in Test house #2
Figure 24. Plenum setup used to evaluate PCO air cleaners
Figure 25. Rust-Oleum Flat black oil-based paint [58] 67
Figure 26. Setup for paint VOC injection
Figure 27. Mr. Clean all-purpose summer citrus [60] 68
Figure 28. Setup for Mr. Clean all-purpose cleaner VOC injection
Figure 29. d-limonene bottle and spray bottle
Figure 30. Setup for pure d-limonene VOC injection70
Figure 31. PCO A core (left) and in plenum (right)72
Figure 32. PCO B core (left) and in plenum (right)73
Figure 33. VOC decay comparison during paint VOC injection experiment
Figure 34. VOC decay comparison during Mr. Clean all-purpose VOC injection experiment 77
Figure 35. VOC decay comparison during d-limonene VOC injection experiment77
Figure 36. VOC linearized decay comparison during paint VOC injection experiment
Figure 37. VOC linearized decay comparison during Mr. Clean all-purpose VOC injection
experiment
Figure 38. VOC linearized decay comparison during d-limonene VOC injection experiment 80
Figure 39. Ozone scrubber
Figure 40. Ozone scrubber attached to ozone monitor
Figure 41. Resetting of the AWAIR IAQ sensors in the living room/dining room area 101
Figure 42. Paint VOC injection – PCO OFF 103
Figure 43. Paint VOC injection – PCO ON – PCO A

Figure 44. Paint VOC injection – PCO ON – PCO B	104
Figure 45. Mr. Clean all-purpose cleaner VOC injection – PCO OFF	104
Figure 46. Mr. Clean all-purpose cleaner VOC injection – PCO ON – PCO A	105
Figure 47. Mr. Clean all-purpose cleaner VOC injection – PCO ON – PCO B	105
Figure 48. d-limonene VOC injection – PCO OFF	106
Figure 49. d-limonene VOC injection – PCO ON – PCO A	106
Figure 50. d-limonene VOC injection – PCO ON – PCO B	107

#### Abstract

#### **RESIDENTIAL INDOOR AIR QUALITY STUDY – ASSESSMENT AND MITIGATION**

Giovanni Josue Cerrato

Thesis Chair: Nelson Fumo, Ph.D.

The University of Texas at Tyler

#### September 2023

An increased awareness in residential indoor air quality (IAQ) has led to an increase in technologies that assist in improving IAQ. Two such technologies include Electro-static (ES) air cleaners and Photocatalytic Oxidizing (PCO) air cleaners. ES air cleaners remove particulate matter from indoor air, but are known to emit ozone, a contaminant, as a by-product. PCO air cleaners abate gaseous pollutants (i.e., Volatile Organic Compounds, VOCs); however, a lack of testing standards creates confusion about their effectiveness in improving IAQ in a real residential home. On the other hand, environmental variables such as temperature and humidity were found to affect VOC sensor readings which complicated the data analysis. In the case of two tested ES air cleaners, the data showed that although the ES air cleaners increased the ozone concentration in the house, the levels are not of concern as they were less than the Food and Drug Administration's (FDA) limit on indoor ozone generation. Regarding two tested PCO air cleaners, minimal indications of an acceleration in VOC abatement were seen. Nevertheless, further insight is given into the methodology behind the real-world testing of PCOs and a correction technique that was used to remove the effect of temperature and humidity on VOC sensor readings.

Keywords: Indoor Air Quality, Ozone, Volatile Organic Compounds, Electro-static Air Cleaner, Photocatalytic Oxidizing Air Cleaner, HVAC

#### PREFACE

Different portions of this thesis are included in publications, from the author, that have published or will be submitted to an academic journal. Those publications are listed in this section for reference and are included with their publication status at the date of this thesis submission.

- G. Cerrato and N. Fumo, "The By-Product of Ozone from Electrostatic Air Cleaners," *American Journal of Undergraduate Research*, vol. 20, no. 1, Mar. 2023.
- G. Cerrato and N. Fumo, "Photocatalytic Oxidizing Air Cleaning Technology A Review" (Manuscript under review)
- G. Cerrato and N. Fumo, "The acceleration of Volatile Organic Compounds by Photocatalytic Oxidizers" (Manuscript under review

#### CHAPTER 1 – INTRODUCTION

#### 1.1 Overview of Indoor Air Quality

Increased awareness of indoor air quality (IAQ), sparked by the COVID-19 pandemic, has led to an increased interest by manufacturers and the public for technologies that assist in improving IAQ [1]. The Environmental Protection Agency (EPA) defines IAQ as "the air quality within and around buildings and structures, especially as it relates to the health and comfort of building occupants [2]." Poor IAQ can lead to a lack of productivity and comfort in addition to negative health effects for indoor residents [3]. The determination of a building's IAQ is dependent on the presence of indoor pollutants which can be measured in concentration levels with the use of IAQ sensors. The primary causes leading to heightened levels of indoor pollution include the collection of indoor sources of air pollutants related to human occupation and the infiltration of outdoor pollution through cracks and openings in a building's envelope. It is reported by the EPA that some air pollutants have indoor concentrations that are two to five times higher than respective outdoor concentrations [4]. The most common indoor pollutants include fine particulate matter (PM), carbon dioxide (CO<sub>2</sub>), volatile organic compounds (VOCs), and ozone (O<sub>3</sub>). Ozone (O<sub>3</sub>) and volatile organic compounds (VOCs) are the contaminants of consideration in this thesis.

#### 1.2 Overview of ozone (O<sub>3</sub>)

Ozone is a highly reactive gas, composed of three oxygen atoms. In the Earth's upper atmosphere, referred to as the stratospheric level, ozone is formed through the interaction of solar radiation with molecular oxygen. Stratospheric ozone is essential as it protects the surface of the earth from harmful ultraviolet (UV) radiation from the sun. Tropospheric ozone is harmful because of its proximity to humans at ground or "breathing" level as an air contaminant. The EPA has an air quality index (AQI) related to 8-hour averages of ozone concentrations, in parts per billion (ppb), separated in categories classified as good (0-54 ppb), moderate (55-70 ppb), unhealthy for sensitive groups (71-85 ppb), unhealthy (86-105 ppb), very unhealthy (106-200 ppb), and hazardous (201+ ppb). The EPA also states that a 2-hour average of 600 ppb ozone concentration is considered a significant harmful level with imminent effects [5]. Tropospheric ozone is considered the outdoor ozone that surrounds buildings and structures. Outdoor ozone is found outdoors as a product of sunlight combining nitrogen oxide (NO<sub>x</sub>) and VOCs generated from automobiles and coal-fired power plants, in addition to VOCs generated from trees and vegetation. This leads to urban areas having higher levels of outdoor ozone than rural areas where there is less pollution. This is also why outdoor ozone is higher in the daytime and, seasonally, in the summertime where there is more sunlight in comparison to wintertime. Geographical location and meteorological conditions are also factors in outdoor ozone concentrations in terms of the production and transport of outdoor ozone [6].

Indoor ozone concentrations will depend on the introduction of outdoor ozone, indoors, through natural or mechanical ventilation and infiltration through the building envelope. Indoor ozone concentrations will also depend on known indoor sources of ozone by some electrical devices that emit ozone as a by-product, such as electrostatic (ES) air cleaners, photocopiers, laser printers, etc. [7]. FDA standard (21CFR801) sets a limit for devices that emit ozone as a by-product, stating that ozone generation should not exceed 50 ppb (parts per billion). This concentration is related to both the volume of air circulating through the device or an accumulation of ozone in an enclosed living space intended to be occupied by humans. Enclosed spaces include houses, apartments, hospitals, and offices [8].

#### **1.3 Overview of VOCs**

VOCs are "organic compounds whose composition makes its plausible for them to evaporate under normal indoor atmospheric conditions of temperature and pressure [9]." The volatility of a VOC depends on its boiling point at standard atmospheric pressure. The lower the boiling point that a VOC has, the more volatile it is classified because it has a greater chance to off-gas from a product or surface in an indoor environment due to it having a higher vapor pressure [9]. For this reason, the World Health Organization places VOCs into three categories: very volatile (VVOCs) due to a lower boiling point, volatile (VOCs), or semi-volatile (SVOCs) due to a higher boiling point [10].

VOCs are found outdoors as a product of vehicle exhaust and burning of fossil fuels, wood, and garbage. VOCs are found indoors as a result of products and materials that contain them, such as building materials (paints, varnishes, caulks, adhesives, carpet, vinyl flooring, composite wood products, and upholstery), home and personal care products (air fresheners, cleaning products, and cosmetics), and human activities (smoking, dry cleaning, cooking, and wood-burning fireplaces) [11]. According to the EPA, the concentration of most VOCs is up to ten times higher indoors than outdoors [12]. In terms of the composition of VOCs in a residential environment, a paper from the Royal Society of Chemistry presents the variability of the different VOCs seen indoor in addition to their respective concentrations indoors. A lot of VOCs correlate with each other, with some VOCs being closely linked to one another in terms of abundance and variability. This means that the presence of a particular VOC may indicate the presence of another. This is the result of a lot of VOCs are mixed together in solvents such as paint or glues, which would explain the previous concept [13].

#### **1.4 Overview of IAQ mitigation**

There are many different technologies that have been developed to reduce the concentrations of pollutants indoors including ventilation strategies, the use of high efficiency filters, and air cleaners/air purifiers. The recommendation of high efficiency filters has become more common in HVAC design guidelines with ASHRAE recommending MERV-13 (or higher) filters in order to reduce the transmission of indoor viruses [1]. However, the use of high efficiency filters leads to increases in air resistance which increases the energy consumption of an HVAC system. The use of ventilation works to address IAQ concerns by diluting pollutant concentrations indoors with cleaner outdoor air. Different ventilation strategies are based on different demand factors such as occupancy, floor size, or actual indoor pollutant levels. When ventilation is limited due to harsh weather conditions or when outdoor air is highly polluted, air cleaners or purifiers can be used. Air cleaners can be sold as portable air cleaners placed in singular locations or can be installed in-duct, working to clean the air in all rooms integrated in the HVAC system. In-duct air cleaners can be installed on the supply side of the air handling unit (AHU) or upstream of the AHU on the return side. In-duct air cleaners work in tandem with the indoor blower and will only purify the air when the blower is powered on.

#### 1.5 Overview of electro-static and photocatalytic oxidizing air purifiers

Two common air cleaners include electrostatic (ES) air cleaners and photocatalytic oxidizer (PCO) air cleaners. ES air cleaners work to reduce PM in an indoor environment by charging dust particles in the air that will be captured by oppositely charged plates. The utilization of high voltages in electrostatic air cleaners to create ionized fields can result in the inadvertent or intentional production of ozone [14]. Air cleaners that intentionally generate ozone are identified as "ozone generators". Producers of such devices erroneously classify ozone as 'activated' or

'energized' oxygen, creating the impression that ozone is a beneficial type of oxygen. The claim made is that the ozone produced from these devices is able to purify the air by removing airborne particles, chemicals, mold, viruses, bacteria, and odors. However, ozone is only moderately effective at cleaning the air at concentrations that are considered unsafe for human exposure. In the case of ES air cleaners that unintentionally produce ozone, they are known do so at much lower levels than ozone generators [14].

PCOs are designed to accelerate the abatement of gaseous, VOC pollutants by way of chemical reactions between a catalyst (usually titanium dioxide) that absorbs gaseous pollutants and a series of UV lights that allow the reactions to take place. The pollutants are either removed or converted into less harmful byproducts such as carbon dioxide and water. The chemical process of reducing VOCs is complex. Some indoor VOCs treated by the PCO air cleaner will not be removed as efficiently as others. In other words, certain PCO air cleaners will react differently with different VOCs [15]. Also, the use of PCO air cleaners can lead to the potential formation of harmful by-products. For this reason PCO air cleaners can be used in tandem with adsorptive filters (i.e. carbon filters) that adsorb gaseous pollutants in addition to filters that contain PCO properties [16] [17] [18] [19]. The EPA's conclusion on PCOs is that their use in homes is limited because currently available catalysts are "ineffective in destroying gaseous pollutants from indoor air. Some PCO cleaners fail to destroy pollutants completely and instead produce new indoor pollutants [20]."

#### 1.6 Thesis outline and purpose

The purpose of this thesis is to contribute to the assessment of indoor air quality and the mitigation of pollutants in residential applications through the examination of two electrostatic (ES) air cleaners and two photocatalytic oxidation (PCO) air cleaners in a real-world application.

The findings from this thesis will lead to the further characterization of the two technologies and their impact on residential IAQ. This will also include assessing the behavior of ozone and VOCs, indoors, to better understand the experimental results. The ES air cleaners were tested to assess their contributions in the increase of indoor ozone levels during their operation and whether they adhered to FDA standards. The PCO air cleaners were tested to assess the extent to which they were able to accelerate the abatement of spikes in indoor VOC concentrations. In terms of the vertical positioning of contaminant measurement, both the ozone and VOC studies were measured at the breathing zone, which is the considered space for human occupancy where the majority of their time is spent indoors. The breathing zone is also defined as, "the area of a room in which occupants breathe as they stand, sit, or lie down" [21]. Measuring pollutants at the breathing zone, as opposed to in-duct measurements, provides a more realistic depiction of occupant exposure and helps prevent overestimation of experimental results.

Additional discoveries from this thesis will support the continued advancement of these technologies through their evaluation in future real-world testing. This will include specifying the design of experimental procedure based on pollutant behavior, the placement of IAQ sensors, and environmental factors affecting VOC sensor readings. A correction model was developed to minimize variations in VOC readings from the sensors that were based on variations in temperature and humidity, not actual changes in VOC concentrations.

Chapter two presents the literature review of ozone, ES air cleaners and their production of ozone, and the evaluation of PCO air cleaners. The ozone review provides insights into the dynamics of indoor and outdoor ozone, enhancing comprehension of the thesis findings that will concentrate on contrasting various baseline levels of indoor and outdoor ozone. The review of ES air cleaners and their production of ozone gives background on the conclusions of previous tests evaluating the technology. Finally, the review on the evaluation of PCO air cleaners narrates through the development of the testing of PCO air cleaners in addition to the identification of relevant design parameters. Chapter three provides the results and discussion of the ozone experiments. Chapter four documents the development of the climatic correction model in reducing variations in sensor VOC readings due to temperature and humidity. Chapter five provides the results and discussion of the VOC experiments. Lastly, Chapter six summarizes the conclusions from the thesis and its relevance to the general theme.

#### CHAPTER 2 – LITERATURE REVIEW

#### 2.1 Literature of ozone and ES air cleaners

#### 2.1.1 Characterization of the tropospheric boundary layer and outdoor ozone

The earth's surface layer is a part of the tropospheric boundary level (BL) which is the lower-level atmosphere that is affected by the friction and transfer of heat from the earth's surface. The tropospheric BL develops periodically throughout the day. In the morning, sunlight heats the earth's surface creating a transfer of heat into the atmosphere; eroding the stable layer formed overnight through radiational cooling. The new BL combines surface heating and wind turbulence and creates what is known as the mixed layer, reaching its maximum depth in the afternoon. The rapid growth of the mixed layer coincides with the mixture of outdoor pollutants [22]. As the sun sets, and solar radiation decreases, a new stable nocturnal BL is established which leads to the discontinuity of pollutants at this level. However, there is a residual layer at a higher elevation that contains pollutants from the daytime. An example of the diurnal cycle can be found in *Jacob, J. D* [23].

Ozone is produced photochemically by the oxidation of methane, carbon monoxide (CO), and non-methane hydrocarbons (NMHCs) in the presence of nitrogen oxides and sunlight. This process leads to ozone having strong diurnals variations near the earth's surface layer. During the afternoon, mixing ratios of outdoor ozone increase at its highest coinciding with the combined effect of photochemical production and the mixing of ozone rich air masses from the residual BL. At nighttime, mixing ratios of outdoor ozone are at its minimum due to dry deposition and a lack of solar radiation in the shallow nocturnal BL [24]. Dry deposition is when particles are removed from the atmosphere due to gravity [25].

#### 2.1.2 Diurnal and seasonal characteristics of outdoor ozone

The diurnal and seasonal cycle of outdoor ozone can be related to solar radiation as stated in the *Introduction*. This correlation is illustrated by a study that was conducted by the Department of Environmental Sciences at King Abdulaziz University (KAU) which observed the diurnal and seasonal variations in Yanbu, Saudi Arabia. Table 1 shows the minimum and maximum outdoor VOC concentrations seen in a diurnal cycle for four seasons, respectively [26].

	Winter	Spring	Summer	Autumn
Maximum VOC				
concentration (ppb) /	28/17.5	39/12.5	36/11.5	32/14.5
Hour of occurrence				
Minimum VOC				
concentration (ppb) /	8/8.5	12/7.5	13/7.5	10/7.5
Hour of occurrence				

Table 1. Maximum and minimum VOC concentrations in a diurnal cycle for four seasons [26]

A couple of things can be concluded from Table 1. One is related to the diurnal cycle of outdoor ozone. The minimum concentrations of ozone occurred at early hours (7:30 or 8:30 AM) for every season. The maximum concentrations of ozone occurred at later hours (11:30 AM, 12:30 PM, 2:20 PM, or 5:30 PM) for every season. This is consistent with the characteristics of the diurnal cycle of outdoor ozone, showed to follow the development of the tropospheric BL mentioned in *2.1.1 Characterization of the tropospheric boundary layer and outdoor ozone*. Outdoor ozone from the residual BL is further mixed and created as solar radiation increases throughout the day. Outdoor ozone levels reach their peak at the height of sunlight. As the sun sets, and solar radiation decreases, outdoor ozone concentrations begin to diminish and stabilize into the nocturnal BL. Another conclusion from Table 1 is that outdoor ozone has a seasonal cycle. The highest range of outdoor ozone concentrations is seen in the spring followed by summer, autumn, and is lowest in

the winter. A ranking of months which can also be related to the amount of solar radiation seen in those months.

#### 2.1.3 Infiltration of outdoor ozone into an indoor environment

Outdoor ozone can infiltrate a building's envelope in three different ways. Through cracks and openings in the exterior, natural ventilation, and mechanical ventilation. A parameter used for measuring the infiltration of outdoor in an indoor environment is the I/O ratio. The I/O ratio relates the concentration of indoor ozone as a percentage of outdoor ozone concentrations. Each of the three paths of infiltration have different I/O ratios. Differences in I/O ratios between the two types of ventilation are because outdoor ozone will pass through more filtering in mechanical ventilation than natural ventilation. Infiltrations through cracks and exterior opening have the smallest effect in raising the I/O ratio and will depend on the tightness of the building's envelope.

A study called "Study of outdoor ozone penetration into buildings through ventilation and infiltration" observed I/O ratios under the three paths of outdoor ozone infiltration. The study included the summary of a literature review on I/O ratio values seen in 385 houses for the three paths of infiltration. I/O ratios were listed as 0.09, 0.19, and 0.47 for infiltration, mechanical ventilation, and natural ventilation respectively. These values corresponded to median air exchange rates, or air changes per hour (ACH), and surface deposition rates. The study concludes listing the paths of infiltration in order from lowest to highest I/O ratios. This means that natural ventilation led to the biggest infiltration of outdoor ozone indoors, followed by mechanical ventilation, and infiltrations through cracks [27].

Another study investigated, "The impacts of building envelope design on indoor ozone and health exposures in residential houses [28]." Data was recorded in four rooms, from three houses with old construction, with different exterior finishes and levels of tightness. None of the houses had any significant sources of indoor ozone such as photocopiers or printers. The houses were occupied with no restriction on their daily routines, and data was taken over the period of 10 days. The constructions of the four envelopes, with respective I/O ratios, are summarized in Table 2.

Envelope	Exterior Wall Finish	Wall surface area (ft <sup>2</sup> )	Window Perimeter (ft)	Window to Wall Ratio	Wall Thickness (in)	I/O ratios
1	Stucco	143	33	0.21	11.0	0.49±0.24
2	Brick	134	33	0.22	16.0	0.52±0.18
3	Brick	176	61	0.30	12.5	0.68±0.19
4	Painted fiber cement siding	79	31	0.95	9.8	0.48±0.2

Table 2. Construction of envelopes 1, 2, 3, and 4 [28]

The factors of the envelopes were statistically analyzed and correlated with indoor ozone concentrations. The study found that the envelope construction variables that played important roles in influencing indoor ozone concentrations were exterior wall finishing and window to wall ratio. Exterior materials can chemically react with outdoor ozone and diffuse ozone before it penetrates indoors. The study reveals mixed effects from the window to wall ratio. The study concluded that the construction variables observed in the study are reasonable predictors of indoor ozone levels.

#### 2.1.4 Generation of ozone from ES air cleaners

In-duct ES air cleaners, installed in HVAC systems, are manufactured to remove a wide range of airborne particles. The standard operating procedure for ES air cleaners consists of three parts which include ionization, collection, and filtering. ES air cleaners first ionize incoming contaminant particles by generating a field of static electricity. The particles are then collected in a series of discharge plates with laminated film envelopes which are separated by a small intermediate distance. The film is a high dielectric material used as an electrical barrier to prevent electric sparks from an electrical discharge [29]. Carbon filters, positioned as the final step in ES air cleaners, are filters that contain granular pieces of carbon. Remaining contaminated particles react chemically with the carbon material and stick to the filter [30]. Thus, preventing the particles from recirculating back into the house. The ionization of the contaminant particles also leads the ionization of oxygen passing through the ES air cleaner. The formation of ozone through ionization can be simplified in a two-step process as seen in Equations 1 and 2 where M is a third stabilizing molecule (M) [31].

$$e + O_2 \xrightarrow{\text{yields}} 2O + e$$
 Equation (1)

$$0 + O_2 + M \xrightarrow{\text{yields}} O_3 + M$$
 Equation (2)

There is a literature review paper named "Electrostatic Precipitators as an Indoor Air Cleaner—A Literature Review" which summarized publications on ES air cleaners. The paper summarizes aspects related to ES air cleaner design that led to higher generation rates of ozone, methods of ES air cleaner testing, and results to the extent that in-duct ES air cleaners had in raising indoor ozone levels in a manufactured test house. The extent to which ozone is generated can be related to product design and operating conditions of the ES air cleaner. Some design factors include charging wire diameter and material, geometry of the ES air cleaner, and the applied voltage. With respect to the amount of voltage applied to the charging wires of the ES air cleaners, a higher operating setting, could lead up to a 50% increase in ozone generation as opposed to a lower setting [32]. A poorly designed ES air cleaner could raise indoor ozone levels above the recommended limit of 50 ppb. For this reason, there are standards given by the California Air Resources Board (CARB) and UL (an accredited standards developer) that are used to certify ES air cleaners in terms of ozone generation with approved test methods [33]. The literature review paper mentioned a study that found the use of the two studied ES air cleaners raised indoors ozone levels by 77 and 20 ppb, respectively. The study concluded that the largest influence in ozone

production were the brands of poorly designed ES air cleaners that generated more ozone than necessary, requiring a high voltage output to the charging wires [34].

Another study, "Characterization of potential indoor sources of ozone", included the observation of indoor ozone levels in homes with permanently installed ES air cleaners. The study first measured indoor background ozone levels downstream of the air cleaner powered off. A total of eight in-duct ES air cleaners were evaluated. The blower ran continuously. The ES air cleaner was powered on, and downstream ozone was measured. The study decided to measure ozone downstream and near supply registers to measure the maximum possible ozone before dissipating into the house. Two of the measured air cleaners produced ozone. However, these measurable quantities of ozone decayed to non-detectable levels as it passed through the supply ducts. The study concludes by stating that the ozone emission rates of the ES air cleaners would not produce concentrations greater than 10 to 30 ppb above background indoor ozone levels [35].

#### 2.2 Literature of VOCs and PCO air cleaner testing

#### 2.2.1 Overview of experimental procedures used in the testing of PCO air cleaners

The testing of PCOs consists of the operating conditions of the PCO and the environment that it is tested in, as well as the specific VOCs that are included in the testing. This also involves the manner in which those selected VOCs are injected into the test environment. VOCs are usually chosen based on either how common they are in an indoor environment according to a chosen reference or their expected reactivity with the PCO air cleaner. The injection methods for VOCs in respective experiments have been either as an evaporated gas to be mixed with air or via diffusion from a surface or liquid into a closed environment. The methodology used in the testing of PCOs has evolved over time in search of being able to simulate the realistic conditions in which the PCO is to be implemented in. Zhong et al. [36] narrate through some of these developments while also providing a systematic assessment of the parameters related to the testing of PCOs. The paper includes a direct observation of the effects that relevant parameters have in the effectiveness of VOC removal by PCO through a series of conducted experiments. The testing performed by Zhong et al. [36] is an improvement in PCO testing in search of a methodology and setup that can simulate the realistic operation of PCOs in HVAC systems. According to Zhong et al. [36], this is in comparison to previous evaluations that have been performed on laboratory bench-top setups under idealistic conditions (low volumetric flowrates, fewer injected VOCs, controlled climates, etc.). The disadvantage of the highly controlled setup is the difficulty in simulating the PCO's performance in full-scale operation. Most PCO studies are performed in controlled laboratory setups, while the need is shifting toward more realistic conditions that the PCO will be applied in. Also mentioned were the differences in objectives found in other evaluations. The majority of studies focus on the design of portable PCO air cleaners in closed chamber testing rather than HVAC applications [37] [38] [39].

The setup used by Zhong et al. [36] was a customized test rig which included a plenum attached to a series of four, parallel aluminum ducts. Inside the ducts were four different combinations of UV lights (ozone and nonozone producing UV lights) and catalysts (titanium dioxide coated on carbon cloth and fiberglass filters). Before VOCs were injected, fans were powered on at a specified airflow rate in addition to powering on the PCO UV lamps. Single VOCs were then injected at a rate deemed appropriate until a desired steady-state VOC concentration was reached. Respective tests for singular VOCs at three different steady-state concentrations were run for an approximate 10 hours. It is important to note that between experiments, the catalysts were regenerated by exposure to UV lamps with a running fan for approximately 10 hours,

removing residual VOCs in the PCO systems from past experiments. The quantification method used in the study was the single-pass removal efficiency of the injected VOC by the PCO. This is characterized by the concentration of the VOC before and after the PCO air cleaner. The relevant parameters affecting PCO removal effectiveness included the type of VOC compound, the VOC concentration entering the inlet of the PCO, the airflow rate, the light intensity of the UV lights, and relative humidity. A summary of the results of Zhong et al.'s testing can be seen in *2.2.2 Laboratory testing of PCO air cleaners*.

The study by Zhong et al. [36] provides a good overall characterization of a methodology that can be used in the testing of in-duct PCO air cleaners while highlighting key parametric factors regarding PCO effectiveness. A broader spectrum is seen in a literature review of different air cleaning technologies performed by Zhang et al. [40] regarding fan-driven, products. Zhang et al. [40] did not conduct a specific individual study but rather gave an overall summary of a number of studies with general commentary. Zhang et al. [40] continue to affirm the lack of systematic assessments on air cleaning technology, especially ones performed under realistic conditions. This also includes a lack of assessments performed long-term. Zhang et al. [40] identified 21 articles deemed to be "relevant and conclusive" in their findings related to PCOs. The review classified the articles into two categories of testing environments (laboratory and field) and two subcategories of laboratory testing methods (single-pass and chamber testing). A single-pass test refers to a method where pollutants are injected in a stream that flows through a chamber connected to the inlet of the air cleaners and exits through a chamber connected to the outlet of the air cleaner. The chamber test method refers to when an air cleaner is placed in a sealed chamber with pollutants injected and mixed within the chamber before the air cleaner is powered on. With respect to PCO testing, Zhang et al. [40] recognized 18 articles related to single-pass laboratory testing, three

related to chamber laboratory testing, and zero articles related to field testing. Zhang et al. [40] go on to say that most studies used only singular VOC compounds which led to "good results". However, because indoor air contains multiple VOC compounds, the results given by those studies could be "misleading". An observation made by Zhang et al. [39] highlighted how the effectiveness of PCO air cleaners drops from single-pass tests to chamber tests which, according to the authors, indicates that "the technology is not ready for practical application."

New attempts in recent years have been made in scaling up laboratory testing to full-scale pilot experiments. One study performed by Shayegan et al. [41] shows the process of scaling up PCO testing using the same pilot experimental setup as Zhong et al. [36], as both papers were published by Concordia University. The study differs from Zhong et al.'s [36] in the sense that it has three experimental setups which include a bench-scale setup, a pilot-scale setup, and a full-scale setup. The bench-scale setup is a linear duct with the PCO being installed within. Compressed air, mixed with injected VOCs, enters the setup and is then subsequentially exhausted. The pilot setup is a plenum attached to four parallel ducts, each of which had a PCO installed. The full-scale setup is an oval-like structure set to mimic an actual HVAC system including a radial fan, a HEPA filter, a cooling coil, etc. The three experiment setups increase in size from the bench-scale setup to the full-scale setup. In addition, the conditions used in each setup become increasingly more realistic of what would be seen in a residential HVAC home (i.e., higher flow rates). Table 3 specifies certain characteristics of the bench, pilot, and full-scale setups. A summary on the results of Shayegan et al.'s [41] testing can be seen in 2.2.2 Laboratory testing of PCO air cleaners.

	Units	Bench-scale	Pilot-scale	Full-scale
Dimensions of PCO (H x L x W)	in	2.8 x 3.5 x 0.3	12.2 x 12.2 x 0.3	24 x 24 x 0.3
Setup volume	$ft^3$	0.22	60	353
Number of UVC lamps	#	2	2	4
Dimension of lamps $(L x D)$	in	2.2 x 0.4	7.9 x 0.6	18.6 x 0.8
Average UVC light intensity	$\frac{W}{m^2}$	36	41	34

*Table 3. Characteristics of the bench, pilot, and full-scale setups [41]* 

Overall, it is clear that the majority of studies performed on PCOs have been in laboratory settings while full-scale experiments are less common. With regard to field testing, the testing or simulation of a PCO in a real-home environment is very limited. The tests that come closest include those that are performed in pilot rooms or test rooms set to replicate an office environment. One European study, conducted by Costarramone et al. [42], included a PCO study with a nonlaboratory chamber test that simulated real life room conditions. The chamber room had air supplied and extracted from it, with VOCs being introduced through the off gassing of a wooden OSB floor and OSB wood panels laid against the room walls. A summary of the results of Costarramone et al.'s [42] testing can be seen in 2.2.3 Field testing of PCO air cleaners. Another study, conducted by Kolarik et al. [43], reviewed the effect that a PCO air cleaner had on perceived air quality in ventilated test rooms. The VOCs injected in the test rooms were emissions from building materials, old ventilation filters, old computer monitors, and human occupation (5 males and 2 females) as a source of human bio effluents. The study differs from most as it evaluated the performance of the PCO in a qualitative manner (according to 34 male and 16 female human subjects) based on odor intensity, air freshness (stuffy or fresh), and air dryness (humid or dry).

These studies are one of the few that approach the real-life application of a PCO air cleaner in a residential house.

#### 2.2.2 Laboratory testing of PCO air cleaners

The experimental results from Zhong et al. [36] revealed a couple of relevant factors in testing in-duct PCOs. The first was the effectiveness of PCOs on different VOC types. The single-pass efficiency for the PCO air cleaners was the highest in the following order of VOC types: alcohols, ketones, aromatics, and alkanes. Single-pass efficiencies specifically for a set of individual VOCs (injected at an initial concentration of 500 ppb) are reported in Table 4. The values are split into three categories of different combinations of PCO UV lights and filters listed as FGF+UVC, FGF+VUV, and CCF+VUV. The two filters included a fiber-glass filter (FGF) and a carbon cloth filter (CCF). The two types of UV lights included ultraviolet (UV) and vacuum ultraviolet (VUV) lamps.

	Single-pass removal efficiency (%)				
VOC	FGF+UVC	FGF+VUV	CCF+VUV		
Ethanol	22.5	26	34		
1-butanol	9	17.5	40		
Hexane	3	13	13.5		
Octane	4	13.5	15		
Acetone	21	21.5	25		
MEK	9.5	12.5	28		
Toluene	13	17	19		
p-Xylene	5	13.25	22		

*Table 4. Removal efficiencies for bench & pilot scale setup [36]* 

The effectiveness of the PCO varies noticeably for different VOCs for a respective PCO. The study highlights the relationship between the intermolecular forces of a specific VOC and the adsorptive ability of the PCO catalyst or filter. Zhong et al. [36] state that, "the present intermolecular forces

are one of the key factors in photocatalytic activity", a point of which can been in the different combination of PCOs. For example, with 1-butanol, the single-pass removal efficiency increases drastically from combination to combination. This is all to affirm that the PCO air cleaner will react differently with different VOCs and that it is the purpose for the development of new catalysts that will react well with a broader range of VOCs. Moving on from Table 3, the second relevant factor taken from the study was the effect that the initial VOC concentration had in the PCO's effectiveness. Focusing on the FGF+UVC PCO, the majority of single-pass removal efficiencies fell for the VOCs as the initial VOC concentrations increased. The observed initial concentrations were 250, 500, and 1000 ppb. The study interprets the behavior as a result of the limited adsorption capacity of the PCO catalyst's surface due to a limited number of fixed active sites. Zhong et al. [36] go on to state that the rate of participating molecules in the PCO reaction is not boosted equally with a noticeable increase in VOC concentration. The third relevant factor was airflow rate. The study reported a gradual decrease in conversion efficiency with an increase in airflow rate. The flowrates observed ranged from 0 to 300 cubic meters per hour (0-177 CFM). An increase in flowrate decreases the residence time, which is the time that VOCs have to be absorbed on the catalyst's surface, therefore giving VOCs less of a chance to be oxidized. The fourth relevant factor was the intensity of irradiance from the UV lights measured in units of energy per area. A gradual increase in VOC conversion efficiency was seen with an increase in irradiance intensity. The final relevant factor, with regard to the UVC PCO, was the effect of humidity on the PCO reaction. In a range of relative humidity (RH) values observed, ranging from 0 to 60% RH, gradual decreases in single-pass removal efficiency were seen with an increase in relative humidity. The study attributes this to the contest between the VOC particles and water molecules for adsorption on the catalyst surface. In other words, humidity in the air acts as a disrupter in the PCO reaction of VOCs. The study also observed the formation of by-products during PCO operation. Zhong et al. [36] go on to state that by-product formation had a close relationship with the PCO's reaction of different VOCs.

The study performed by Shayegan et al. [41] goes even further in showing the development of more realistic testing to demonstrate the effectiveness of PCO air cleaners. Shayegan et al. [41] compare the different setups in a specific section and do so by keeping the airflow velocity constant. The bench-scale setup is compared to the full-scale at 1.25 m/s (4.1 ft/s), while the pilotscale setup is compared to the bench-scale setup at 0.5 m/s (1.6 ft/s). The VOCs chosen were toluene and isobutanol at initial concentrations of 100 and 1,000 ppb. Table 5 and Table 6 show the approximate results for the UVC PCO in the bench & pilot scale setups and bench & full-scale setups, respectively.

	Removal efficiency (%) at 4.1 ft/s of air flowrate				
VOC	Be	Bench		ilot	
	100 ppb	1,000 ppb	100 ppb	1,000 ppb	
Toluene	16	7.6	7.8	6.6	
Isobutanol	70	44	56	25	

Table 5. Removal efficiencies for bench & pilot scale setup [41]

 Table 6. Removal efficiencies for bench & full-scale setup [41]

	Remo	Removal efficiency (%) at 1.6 ft/s of air flowrate			
	Bench		Full		
VOC	100 ppb	1,000 ppb	100 ppb	1,000 ppb	
Toluene	11	3	0	0	
Isobutanol	55	26	24	8	

A lot of trends seen in comparing the different scaled experiments are related to the relevant factors of PCO testing discussed by Zhong et al. [36]. In setups comparing two concentrations of

the same VOC, the removal efficiency drops due to the limited adsorption capacity of the PCO. When comparing the bench setup at different airflow velocities and the pilot and full-scales setups (operating at different velocities), the reduction in removal efficiencies resulting from increases in velocity are due to less resident time of the contaminated air under the UV lights. The size of the system is another factor that is seen when comparing the bench setup to the pilot and full-scale setups working under the same airflow rate. As seen in Table 5 and 6, an increase in size led to a decrease in removal efficiency which can be linked to differences in light distribution, most uniform in the bench-scale setup. Lastly, the differences seen in the removal efficiencies of the VOCs are related to the respective reaction of each VOC with the PCO. In short, Shayegan et al. [41] conclude their study by stating that decreases in removal efficiency can be attributed to the size of the system, the inlet VOC concentration, and the air velocity passing through the PCO. Shayegan et al. also stated that "that small-scale results are not sufficiently accurate and reliable for direct extrapolation to large-scale application in the design of the UVC-PCO system." The authors add that, "further studies are required to evaluate the effect of scaling up for indoor real conditions..."

In Zhang et al.'s [40] literature review, they conclude that PCO air cleaners can reduce the concentrations of some VOCs. However, they can produce harmful by-products too. The authors add that standard conditions and procedures are needed for the evaluation of air cleaner performance. Another literature review was performed by Mata et al. [44] reviewing air cleaning technologies in general. The study focuses on the selection of articles that present technologies that are either ready or close to being applied in indoor environments by consumers. One article presented in the literature review by Mata et al. [44] was by Weon et al. [45] which presented a PCO air cleaner design that was evaluated in a chamber test. The article focused on the

improvement of a titanium-dioxide catalyst that is more efficient in the scaled-up application of a PCO. Five VOC gases were injected at a target concentration of 10,000 ppb with each VOC being degraded at different rates, respectively. The article notes that the order of removal efficiency from highest to lowest was formaldehyde, ammonia, acetaldehyde, acetic acid, and toluene, respectively. The study also states that the PCO air cleaner with the specified photocatalyst filter was able to achieve 72.1% average removal efficiency in 30 minutes of operation. Testing protocols were based on a standard by Korea Conformity Laboratories (KCL). The findings from Weon show major promise in the development of a PCO air cleaner to be used in indoor applications with results showing high removal efficiency with high initial concentrations. It is important to note that the article did not mention the specific airflow rate or velocity but rather refers the reader to the KCL standard used. Referring back to the literature review by Mata et al. [44], the group concludes by saying that PCO air cleaners are promising air cleaning technology but there are aspects that need to be considered before their use in indoor air applications. They continue to affirm the factors discussed in this paper (i.e., light intensity, air flowrate, concentration levels, relative humidity, etc.) as the relevant variables in PCO design. By-products continue to be an issue with PCO efficacy.

#### 2.2.3 Field testing of PCO air cleaners

Now, the focus will shift towards highlighting a couple of studies that have been performed in non-laboratory experimental setups and what they have to say about PCO effectiveness. Even though the qualitative-perceived air quality study performed by Kolarik et al. [43] does not present a quantitative evaluation of the PCO in terms of concentration, it is worth mentioning that the application of the PCO in the study was close to a real-life application of a PCO. The conclusions of the study were that human occupants perceived improvements in the air quality of the test rooms polluted by the building materials, used ventilation filters, and old computer monitors, but not of test rooms polluted solely by human emissions (consistent of alcohol VOC compounds). The study attributes an increase in dissatisfaction of perceived air quality to incomplete oxidation of the most common alcohol VOC types emitted off humans, in addition to the creation of by-products from the PCO. The study included a numeric value, called Clean Air Delivery Rate (CADR) which measures the air-cleaning effect of the PCO air cleaner. The computation of CADR is used and explained in *5.6 Evaluation of PCO performance*. The perceived effectiveness expressed as CADR depended on the different types of pollution sources which can be linked to different VOCs, altogether. An observation in *2.2.2 Laboratory testing of PCO air cleaners* related to differences in removal efficiency due to different VOC compounds and types.

The European pilot room study, performed by Costarramone et al. [42], does present the quantitative evaluation of a PCO installed in a non-laboratory setting. The study included four commercial PCO air cleaners bought without adding any modifications. Two pilot rooms were considered, with one being thermoregulated (labeled the BEF platform) and the other not being thermoregulated (labeled the EVALIS platform). The two PCO air cleaners (named E8 and E15) labeled as the most efficient, as per a set of separate laboratory testing, were placed in the BEF pilot room regulated at 77 °F and 60-80 %RH. Tables 7 and 8 show the approximate variations in initial VOC concentration (converted to ppb) seen from the powering on of the E8 (for 2 and 5 days) and E15 (for 3.5 days) PCO air cleaners, respectively. Positive values are representative of percent increases while negative values are representative of percent reductions in VOC concentration.
	Initial concentration (ppb)	Variation of co	oncentration (%)
VOC	0 days	2 days	5 days
Formaldehyde	9.8	96	88
Acetaldehyde	8.2	36	58
Acetone	43	-8	4
Toluene	4	-4	-72
Limonene	9.3	-46	-70
Hexanal	26.4	-58	-76
Alpha-pinene	45.2	-50	-62
Beta-pinene	11.1	-56	-70
Pentanal	6.3	-58	-78
1-pentanol	3.3	-56	-80
Octanol	5.45	-60	-82
3-carene	10.4	-54	-70
nonanal	6.5	-60	-80

Table 7. Variations in VOC concentration for the powering on of the E8 PCO [42]

Table 8. Variations in VOC concentration for the powering on of the E15 PCO [42]

	Initial concentration (ppb)	Variation of concentration (%)
VOC	0 days	3.5 days
Formaldehyde	12.2	20
Acetaldehyde	11.4	-8
Acetone	46	-24
Toluene	2.4	-56
Limonene	11	-78
Hexanal	34.2	-80
Alpha-pinene	55.3	-76
Beta-pinene	12.4	-78
Pentanal	7.7	-80
1-pentanol	5	-88
Octanol	7.5	-72
3-carene	11.9	-76
nonanal	8.26	-88

Not shown in the data given in Tables 7 and 8 is that the study reports that the E15 did not release any by-products during its operation while the E8 was shown to generate  $NO_x$  as a by-product. The results show a couple of trends regarding the effectiveness of the evaluated PCOs. First, the PCOs were most effective in reducing the heaviest VOCs. On the other hand, the concentrations of Formaldehyde, Acetaldehyde, and Acetone, which are considered light VOCs, showed minimal changes or even increased. Also noticeable was the rate at which the concentration dropped. As seen in Table 6, certain VOCs reached higher percentages of VOC reduction at 2 days than other VOCs. Toluene had the slowest reduction as its percent reduction was 4% on day 2 before exponentially increasing to 72% in day 5. Overall, the comparison of 2 to 5 days for the E8 PCO air cleaners showed that longer operation led to continual degradation of the VOCs. It is important to note that while the majority of VOCs experienced reductions in concentrations, the initial concentrations were low in magnitude, which bodes well for the PCO's performance. Also not seen in the tables was information the study gave on the aging of the PCO's photocatalyst. The study mentions that in a long-term testing of the PCO, the quantitative clean air delivery rate (CADR) reached a high after a month, remaining stable after two months. CADR decreases at a faster pace after the duration of 4 months. The study attributes this to an aging in the output of the UV lamps and saturation of the photocatalytic media. The results of Costarramone et al.'s [42] testing give valuable insight in the testing of PCO air cleaners under conditions close to real-world applications.

As stated in the section 2.2.1 Overview of experimental procedures used in the testing of *PCO air cleaners*, the testing of PCO air cleaners in the field is extremely limited. This may be due to the factors discussed in the section 2.2.2 Laboratory testing of PCO air cleaners. Since the laboratory testing of PCO air cleaners has resulted in mixed results, the focus of PCO research has

been on the development of new photocatalysts that are better suited to be tested in scaled-up, realistic environments.

# CHAPTER 3 – METHODOLOGY AND RESULTS FOR THE EVALUATION OF OZONE PRODUCTION FROM TESTED ELECTRO-STATIC AIR CLEANERS

#### **3.1** Testing facilities for ozone experiments

The ozone experiments were conducted at the two test and research houses at the University of Texas at Tyler. The test houses are identical in size and layout with an area schedule of 1,470 square feet (excluding the garage, covered porch, and covered patio). The houses have an interior volume of 11,939 cubic feet and are shown in Figure 1. Test house #2 (on the left) has a tight building envelope (representative of new construction methods) while test house #1 (on the right) is draftier (representative of older construction). Both test houses have infiltration ratings quantified as air changes per hour ACH at 50 pascals (ACH50) from a blower door test reported in the air leakage reports of the houses. Test house #1 and #2 have ACH50 ratings of 4.47 and 1.87, respectively.

They both have a brick exterior finish with a total of nine windows for each house. The houses each consist of 3 bedrooms, 1.5 bathrooms, and an open concept area which includes the living room, dining room area, and kitchen. Neither test houses have regular occupants, nor are they furnished.



Figure 1. Test houses at the University of Texas at Tyler

## **3.2 Ozone monitoring**

The model 202 ozone monitor from 2B Technologies was used to monitor ozone in the living room. The ozone monitor was zeroed out for every experiment (see *Appendix A*). Readings from the ozone monitor have an uncertainty of  $\pm 1$  ppb or 2% of the reading. Since all readings inside the house are below 50 ppb, the uncertainty associate of readings inside the house is 1 ppb. The ozone monitor was stationed inside the test house and placed in two different positions throughout all the ozone experiments. For ozone experiments 1 through 4, the ozone monitor was placed on a stool underneath the supply registers as illustrated in Figure 2. In ozone experiment 5, the ozone monitor was placed on a ladder about an inch and a half from the ceiling close to the supply registers as illustrated in Figure 3. The location of the stool setup was determined as the result of a stratification test and a test that measured the critical point of ozone in the test house (*see Appendix B*). Outdoor ozone data was received from the environmental monitoring station in Tyler, Texas, recorded in the Texas Commission on Environmental Quality (TCEQ) database [46]. The uncertainty of the outdoor ozone readings given by the ozone monitoring station have an uncertainty of 0.5 ppb.



Figure 2. Stool setup for the ozone monitor



Figure 3. Ladder setup for the ozone monitor

Figure 4 shows the location of the ozone monitor and the four nearest supply registers with respect to the house layout.



Figure 4. Ozone monitor location and nearest supply registers in house layout

# 3.3 Overview of ozone experiments

All ozone experiments included the AHU fan being powered on continuously at 100% corresponding to 803 CFM, which was given by the diagnostics portal of the HVAC system. Setting the AHU fan on at 100% allowed for good air mixing within the test house at 4.03 air changes per hour (ACH). Experiment 1 was set to show baseline indoor ozone levels (due to infiltrations only) in house #2. Experiment 2 compared the baseline indoor ozone concentrations of the test and research houses. Experiment 3 highlighted the increase in indoor ozone from

experiment 1 due to ventilation, and in experiment 4, the ES air cleaner is powered on. The amount of mechanical ventilation used in ozone experiments 3 and 4 was 6.3% of fresh air intake. This percentage was computed by dividing the flowrate of fresh air introduced into the house (50 CFM) by the supply flowrate (803 CFM). In experiment 5, ventilation was powered off to show the increase in indoor ozone due only to the powering on of an ES air cleaner. Experimental 6 is identical to experiment 5 with the only difference being that indoor ozone concentrations were measured near the supply registers. Experiments 3 through 5 were repeated for both ES air cleaners A and B.

Certain technical details of the ES air cleaners including the dimensions, material, and specific voltage generated from the charging wires are not available publicly as they are not listed in the technical specifications of the air cleaners. A disclaimer of which should be taken into consideration when interpreting the results of the study. There is a documented difference in the operation setting of the ES air cleaners which is the voltage level (maximum, medium, or minimum) applied to their respective charging wires. ES air cleaner A had an actual distinction between "on" and "max-on", while ES air cleaner B only had an "on" option.

Table 9 summarizes the setup of all six ozone experiments.

Experiment	Mechanical ventilation	Sensor location	House	Duration (Days)
1	Off	On stool in living room	2	1
2	Off		1 and 2	1
3	On			ES air cleaner A: 1 ES air cleaner B: 1
4	On		2	ES air cleaner A: 1 ES air cleaner B: 1
5	Off		2	ES air cleaner A: 2 ES air cleaner B: 3
6	Off	On ladder in living room		ES air cleaner A: 2 ES air cleaner B: 3

Table 9. A summary of the setup for all six ozone experiments

### 3.4 Results and discussion of ozone experiments

For all plots in the ozone results section, the horizontal green line represents the FDA standard 21CFR801 mentioned in the *introduction* and the vertical red line separates days of data.

# 3.4.1 Baseline indoor ozone levels in UTT house #2 (Ozone experiment 1)

Figure 5 shows the ozone data for ozone experiment 1 which was performed in test house 2 with the sensor located on a stool in the living room, without mechanical ventilation.



Figure 5. Ozone experiment 1 results for test house #2

Since there is no source of ozone inside the house with the electrostatic filter off, and no ventilation, levels of ozone can only be affected by infiltration. As discussed in 2.1.5 Infiltration of outdoor ozone into an indoors environment, I/O ratios tend to be affected the least due to infiltrations and will depend on the tightness of the building's envelope. This seems to be the case from the data in Figure 5. The indoor ozone levels are following the outdoor trend at a very low magnitude with a moderate correlation coefficient of  $R^2 = 0.47$ . This is representative of the fact that house #2 is constructed with a tight envelope. The average baseline concentration of indoor ozone seen in the house was  $2.7\pm1$  ppb, corresponding to a I/O ratio of  $0.08\pm0.03$ , which will be

used as a reference for comparison in determining the increase in indoor ozone when the ES air cleaner is powered on.

3.4.2 Comparison of baseline indoor ozone levels in the UTT test houses (Ozone experiment 2)

Ozone experiment 2 was performed in both test houses 1 and 2 with the sensor located on a stool in the living room of each house respectively, without the use of mechanical ventilation. Figure 6 shows the baseline indoor levels of ozone in Test house 1 and 2 for a period of 24 hours where average outdoor ozone concentrations were closest.



*Figure 6. Ozone experiment 2 results for test houses #1 and #2* 

Both houses are identical in layout, window to wall ratio, and size (factors related to infiltrations in a house envelope discussed in 2.1.5 Infiltration of outdoor ozone into an indoor environment) with the only difference being that house #2 has a tighter envelope than house #1. Therefore, it can be expected that house #2 will have lower indoor ozone levels than house #1. As seen in the data from Figure 6, the indoor ozone levels in house #2 followed outdoor levels at a low concentration (around an average of  $3.6\pm1$  ppb), while the indoor levels in house #1 had a higher concentration (around an average of  $8.6\pm1$  ppb).

Figure 7 shows normalized results of both houses in continuity, where the vertical black line separates house #2 (hours 0-48) and house #1 (hours 48-168). The indoor and outdoor ozone levels are normalized in a range from zero to one in percentage of their highest value seen from house #1 and house #2. Figure 7 also includes the I/O ratios of both houses in continuity.



Figure 7. Normalized baseline indoor ozone levels for test house #1 (0-48 hours) and house #2 (48-168 hours) Figure 7 provides a different visual for the same trends illustrated in Figure 6. House #2 is more air-tight, so the indoor ozone levels follow the outdoor ozone levels at a lower concentration than after the 48-hour mark when the ozone monitor was moved to test house #1. This is also represented by the increase in I/O ratio values after the sensor was moved to house #1. The results of ozone experiment 2 further highlight the tightness of house #2, with respect to the infiltration of outdoor ozone indoors which was an observation already seen with the results of ozone experiment 1.

3.4.3 Indoor ozone levels with mechanical ventilation in UTT house #2 (Ozone experiment

3)

Ozone experiment 3 was performed in test house 2 with the sensor located on a stool in the living room, and with the use of mechanical ventilation. Figures 8 and 9 shows the results of ozone experiment 3 for both ES air cleaners.



Figure 8. Ozone experiment 3 results during ES air cleaner A's testing



Figure 9. Ozone experiment 3 results during ES air cleaner B's testing

As discussed in 2.1.5 *Infiltration of outdoor ozone into an indoor environment*, infiltration through mechanical ventilation and natural ventilation (through the opening of windows and doors) leads

to higher I/O ratios. Ozone experiment 3 observes indoor ozone levels in test house #2 based on the influence of mechanical ventilation. With intake and exhaust ventilators being powered on, outdoor air is being directly introduced into the test house which has higher levels of ozone that is being mixed with indoor air. This causes the indoor ozone level trends to follow outdoor ozone level trends more closely, as seen in the plots. Quantitatively, the two tests included correlation coefficients of  $R^2 = 0.52$  and  $R^2 = 0.61$  between indoor and outdoor ozone. These coefficients are both higher that the coefficient in ozone experiment one (baseline ozone levels)  $R^2 = 0.47$ . In terms of magnitude, the I/O ratio increases from the baseline indoor ozone levels. The I/O ratios in ozone experiment 3 for ES cleaners A and B were 0.13±0.4 and 0.17±0.05, respectively. These ratios are both increases from ozone experiment 1 which had an I/O ratio of 0.08±0.03. In terms of observing the plots, one can see higher peaks (hour 16 in Figure 8 and hours 0 and 24 in Figure 9) of indoor ozone levels in ozone experiment 3 as opposed to the peak in indoor ozone levels (hour 17 in Figure 5) seen in ozone experiment 1. The results of ozone experiment 3 are later compared to the results of ozone experiment 4 to observe increases in indoor ozone levels due to the powering on of the ES air cleaners during a period with mechanical ventilation.

3.4.4 Indoor ozone levels with mechanical ventilation and the powering on of two ES air cleaners in UTT test house #2 (Ozone experiment 4)

Ozone experiment 4 was performed in test house 2 with the sensor located on a stool in the living room, and with the use of mechanical ventilation. The ES cleaners were powered on. Figures 10 and 11 show the indoor ozone data for ozone experiment 4, which had durations of 24 hours for ES air cleaners A and B, respectively.



Figure 10. Ozone experiment 4 results during ES air cleaner A's testing



Figure 11. Ozone experiment 4 results during ES air cleaner B's testing

In ozone experiment 4, outdoor ozone is still being directly introduced into the test house through mechanical ventilation. There is such an indication when comparing the plots of ozone experiments 3 and 4 for ES air cleaner A (Figures 8 and 10). Even though lower outdoor ozone levels are seen in ozone experiment 4 in comparison to experiment 3, the indoor ozone levels in the house were higher. This distinction was also seen in the testing of ES air cleaner B (Figures 9 and 11). Quantitatively, comparing ozone experiments 3 and 4, increases in indoor ozone levels were seen of  $2.6\pm 2$  ppb during the testing of ES air cleaner A and  $5\pm 2$  ppb during the testing of

ES air cleaner B. Changes in outdoor ozone levels included a decrease of  $0.2\pm9$  ppb during the testing of ES air cleaner A and an increase of  $4.2\pm9$  ppb during the testing of ES air cleaner B. The I/O ratios in ozone experiment 4 for ES cleaners A and B were  $0.21\pm0.05$  and  $0.29\pm0.05$ , respectively. These ratios are both increases from the I/O ratios in ozone experiment 3 which were  $0.13\pm0.04$  and  $0.17\pm0.05$  for ES air cleaners A and B, respectively. In both cases there is an indication of increases in indoor ozone due to the powering of the ES static air cleaners. However, it is uncertain what percentage of the increase was due to the ES air cleaners or the influence of mechanical ventilation. The comparison shows that neither ES air cleaner, in combination with mechanical ventilation, exceeded the FDA limit for ozone levels.

3.4.5 Indoor ozone levels with the powering on of ES air cleaners in UTT test house #2 (Ozone experiment 5 and 6)

Ozone experiment 5 was performed in test house 2 with the sensor located on a stool in the living room, without the use of mechanical ventilation. The ES cleaners were powered on. The results from ozone experiment 5 are shown in Figures 12 and 13 for ES air cleaners A and B, respectively.



Figure 12. Ozone experiment 5 results during ES air cleaner A's testing



Figure 13. Ozone experiment 5 results during ES air cleaner B's testing

Ozone experiment 5 allows for a more direct evaluation of the contribution that the ES air cleaners had in raising indoor ozone levels. In experiment 5, the only indoor sources of ozone are the ES air cleaners with indoor ozone levels being influenced by outdoor levels solely through infiltrations. This allows for a distinct comparison between the indoor levels from ozone experiment 1 and experiment 5. For ES air cleaner A, the inside ozone levels varied around an average of 10.5±1 ppb (Figure 12) where in ozone experiment 1 they varied around an average of 2.7±1 ppb (Figure 5). ES air cleaner A, on max setting, increased the ambient ozone levels in the house by approximately  $7\pm 2$  ppb. ES air cleaner B increased the indoor ozone concentration only minimally from baseline ozone levels. ES air cleaner B contributed to indoor ozone levels that varied around an average of  $4.2\pm1$  ppb (Figure 13) leading to an approximate increase of approximately 1.5±2 ppb. Decreases in outdoor ozone levels between experiments were 3±3 ppb during the testing of ES air cleaner A and  $13.8\pm4$  ppb during the testing of ES air cleaner B. The I/O ratios in ozone experiment 5 for ES cleaners A and B were 0.33±0.05 and 0.2±0.08, respectively. These ratios are both increases from ozone experiment 1 which had an I/O ratio of  $0.08\pm0.03$ .

As stated in 2.1.6 Generation of ozone from ES air cleaners, ES air cleaners ionize incoming contaminant particles by generating a field of static electricity. In search of further reducing contaminant particles, different ES air cleaners will generate higher fields of static electricity based on respective design factors and/or higher operation options. Since ES air cleaner A has a "maximum-on" operation option while ES air cleaner B has an "on" option only, it is expected that ES air will generate more ozone due to a higher operation level of ionization. It is also important to note that the increases seen with ES air cleaner A and B were at the lower end of the 10-77 ppb increases mentioned in 2.1.6 Generation of ozone from ES air cleaners.

Both air cleaners contributed to an increase in indoor ozone concentrations even with decreases in outdoor ozone levels. The statistical significance of the results was found with a hypothesis test for the difference of means. This included the computation of a p-value which describes the probability of the test reaching the null hypothesis, which is described below. The averages of indoor ozone for experiment 5 of both ES air cleaners (where the ES air cleaners were the only source of indoor ozone) were compared to the average of indoor ozone in experiment 1 (baseline ozone levels). In order to compare the ES air cleaners together, data was taken from the first 48 hours of ES air cleaner A's experiment 5 to match the sample size of ES air cleaner B's experiment 5. The null hypothesis was that the ES air cleaners did not contribute to higher ozone levels than baseline ozone levels. Both ES air cleaners had p-values of less than 0.00001 when compared to baseline indoor ozone levels, which leads to the data results being statistically significant to a significance level of 95%. This led to the conclusion that the average indoor ozone concentration was actually higher when the ES air cleaners were powered on and that it was not due to chance. Nevertheless, the increase in indoor ozone from either of the ES air cleaners did not exceed the FDA 21CFR801 standard of 50 ppb mentioned in the Introduction.

Ozone experiment 6 was performed in test house 2 with the sensor located on a ladder in the living room, without the use of mechanical ventilation. The ES cleaners were powered on. The results of the final ozone experiment, experiment 6, are shown in Figures 14 and 15 for ES air cleaners A and B respectively.



Figure 14. Ozone experiment 6 results during ES air cleaner A's testing



*Figure 15. Ozone experiment 6 results during ES air cleaner B's testing* 

Ozone experiment 6 focuses on the differences in indoor ozone levels observed near the supply registers than when observed in the living space. The idea is that since the ES air cleaners are producing a by-product of ozone, ozone levels should be higher near the supply registers before

diluting into the air of the living space. However, this did not seem to be the case. Changes in indoor ozone levels between ozone experiments 5 to 6 included a decrease of  $2.2\pm2$  pbb during the testing of ES air cleaner A and a slight increase of  $0.3\pm2$  pbb during the testing of ES air cleaner B. Increases in outdoor ozone levels between experiments were  $2\pm4$  pbb during the testing of ES air cleaner A and  $9.9\pm4$  pbb during the testing of ES air cleaner B. The I/O ratios in ozone experiment 6 for ES cleaners A and B were  $0.25\pm0.04$  and  $0.15\pm0.04$ , respectively. These ratios are both decreases from the I/O ratios in ozone experiment 5 which were  $0.33\pm0.05$  and  $0.20\pm0.08$  for ES air cleaners A and B, respectively. Referring back to the plots, the indoor ozone for both ES air cleaners (Figures 14 and 15) seemed to remain very close to the indoor levels in ozone experiment 5 (Figures 12 and 13), when the sensor was on a stool. This concept is re-enforced in the stratification of ozone test discussed in *Appendix B*.

#### **3.5** Conclusions from ozone experiments

Both ES air cleaners contributed to increases in indoor ozone levels in test house #2. ES air cleaner A showed an increase in ozone magnitude from 2.7 to 10.5 ppb (approx.  $7\pm2$  ppb), while ES air cleaner showed a minimal increase from 2.7 ppb to 4.2 ppb (approx.  $1\pm2$  ppb). These results were statistically significant with very low p-values. In terms of increases in I/O, ES air cleaner A showed an increase of  $0.25\pm0.02$ , while ES air cleaner B showed an increase of  $0.12\pm0.05$ . Even though the ES air cleaners were found to give off a by-product of ozone, neither of the ES air cleaners surpassed the indoor limit of ozone based on the FDA standard 21CFR801 with ES air cleaners A and B being 39.5 and 45.8 ppb below the 50-ppb ozone limit, respectively. It is also noteworthy that the indoor ozone levels observed at the supply registers were not noticeably higher than ozone levels measured at the living space, confirming the stratification test in *Appendix b*. The ES air cleaners also increased indoor ozone levels during a period where test

house #2 was being mechanically ventilated. Nevertheless, neither of the ES air cleaners, in combination with mechanical ventilation, raised indoor ozone levels above the FDA limit with ES air cleaners A and B being 43.4 and 39.4 ppb below the 50-ppb ozone limit, respectively.

These findings contribute to determining the safety of using ES air cleaning technology in residential applications. Thus, providing information that will assist consumers in making decisions on IAQ mitigation strategies and help manufacturers in the future development of ES air cleaning technology in residential applications. Additional findings include that the powering on of ventilation led to the indoor ozone levels being more correlated to outdoor ozone, and that indoor ozone levels seen in test house # 2 were lower than those seen in test house #1, which had the draftier insulation.

# CHAPTER 4 – CORRECTION MODEL FOR CLIMATIC VARIABLES AFFECTING VOC READINGS

#### 4.1 Overview of multi-pixel gas sensors

In this thesis, Element AWAIR sensors were used to measure VOC concentrations in the VOC experiments performed at the UTT test homes. The AWAIR element sensors use a multipixel metal oxide as their respective sensing element. In a review of low-cost sensors for indoor air quality, Garcia et al. [47] discuss the technology and components used in sensors to read VOCs which includes metal oxide semiconductors or MOS [47]. Garcia et al. [47] state that based on laboratory and on-field studies, it is known that MOS sensor sensitivity varies due to temperature and humidity. Another paper written by Abdullah et al. [48] is even more specific [48]. Abdullah et al. [48] write that the method for which the MOS sensors read VOCs is through the adsorption of VOCs on the device's sensing material. Temperature and humidity will both affect the adsorption of VOCs onto the sensor's sensing material. With an increase in temperature, there is an increased adsorption of oxygen on the sensing material which causes a decrease in sensitivity in the sensor's reading. An increase in humidity will cause water vapor to adsorb onto the sensing material causing a "driff" in the sensor's readings.

According to testing done by Abdullah et al. [48], observing sensor responses due to temperature and humidity, it was found that the two environmental variables are inversely related with sensor resistance. Achieving a steady-state VOC concentration, increases in temperature and humidity led to decreases in sensor resistance while decreases in temperature and humidity lead to an increase in sensor resistance. A decrease in sensor resistance leads to higher VOC concentration readings that is unrelated to any injections of VOCs in the test chamber while an increase in sensor resistance resistance would lead to the opposite effect [49]. Therefore, with this sensor technology, variations

in temperature and humidity will lead to variations in VOC readings that are unrelated to actual changes in VOC concentrations.

#### 4.2 Effect of temperature and humidity on VOC sensor readings

A baseline was run at the test house in a period of time where no VOCs were injected into the UTT test home. During this period, the thermostat was manipulated to cool and subsequentially heat the house to observe changes in sensor VOC readings due to changes in temperature and humidity. Figure 16 shows the results of the baseline with labels for when the thermostat was manipulated.



Figure 16. Variations in VOC concentrations due to temperature and humidity

Ideally, it is expected that since no VOCs are injected into the house during this period of time, the VOCs would vary minimally and remain close to a constant. However, it can be seen in Figure 16 that as the thermostat is set on cooling mode, the VOC concentrations immediately start to decrease as both climatic variables start to decrease. When the thermostat is changed to heating mode, an immediate spike is seen in humidity leading to a spike in VOC concentrations even though no VOCs were injected. This baseline reveals a couple of things regarding the effect of the climatic variables on the VOC sensor readings. One is that variations in temperature and humidity

cause immediate variations in VOC sensor readings that are unrelated to actual changes in VOC concentrations. The second is that variations in humidity have a greater effect on variations in VOC concentrations than temperature. This can be seen at hour 19.5 where there was a spike in VOC concentrations corresponding to the spike in humidity as opposed to a gradual increase as seen with temperature. Specifically, in the VOC spike between hour 19 and 19.83, a 90% increase in VOC concentration was seen corresponding to increases in temperature and humidity of 6% and 82%, respectively.

In the main VOC experiments, the compressor of the HVAC was powered off which means the house was not thermoregulated. Indoor temperature and humidity varied naturally based on the outdoor conditions, mainly due to infiltration. Overall, as can be seen with the information from *4.1 Overview on multi-pixel sensors* and the baseline presented in this section, the variations in VOC concentrations due to the climatic variables were required to be corrected in order to see the true trends in VOC concentrations.

#### 4.3 Methodology for VOC correction model

To combine the effects of temperature and humidity into a single variable, the total enthalpy of the air was used to represent the complex interactions of the climatic variables for the development of the correction model. The formula used for total enthalpy of the air is shown in Equation (3) from the ASHRAE fundamentals handbook IP 2017 where *T* is temperature in °F and  $\omega$  is the humidity ratio [50]. These enthalpy values are associated to each individual reading of VOC concentration to be used for the implementation of the model.

$$h = 0.24 + \omega(1061 + 0.444T)$$
 Equation (3)

The correction model works to observe variations in VOC concentrations due to variations in enthalpy and remove those effects. The results of each VOC experiment are a corresponding decay from a spike in VOC concentration due to the injection of VOCs. The effects of enthalpy on the decay were evident as "humps" in VOC concentration were seen. One of the results of the VOC experiments is shown in Figure 17 to illustrate the previously stated behavior. The final plots of uncorrected and corrected experimental data are in *Results and discussion of VOC experiment*.

To determine the specific effect that enthalpy had on each experiment, a period of the data at the end of the decay that included a "hump" was taken (outlined in a red box in Figure 17) where the VOC concentration is expected to be constant (i.e., is not expected to change).



Figure 17. Uncorrected VOC data

During this period of time (outlined by the red box), VOC concentrations are plotted against enthalpy values. A linear trendline is then computed to obtain the slope (M) which characterizes the change in VOC concentrations according to enthalpy as seen in Figure 18.



Figure 18. Calculation of slope to be used in correction model procedure

According to the corresponding variation in enthalpy of every VOC concentration reading, the slope was used as a correction coefficient. Variations in enthalpy from the main data set were computed by calculating the difference in every value of enthalpy and the initial value of enthalpy. The product of the calculated slope to the calculated variation in enthalpy corresponding to each data point is characterized as the correction factor for every data point (denoted as  $F_n$ ). To obtain the new corrected VOC concentration, the correction factor ( $F_n$ ) for every data point is subtracted from the original reading from the sensor. It is important to note that not all data points were selected to be inputted into the correction model. The data selected included the initial decay of VOCs, where no "humps" in the data were seen, in addition to the lowest points in the "dips" seen in the original data set. The overall procedure of the correction model is summarized in Table 9.

Step	Implementation
1. Obtain the slope ( <i>M</i> ) characterizing the change in VOC concentration to enthalpy.	<ul> <li>Take the period of the data at the end of the decay that includes the last "hump," where the VOC concentration is expected to be constant.</li> <li>Plot VOC concentration vs enthalpy during that period to get the slope (<i>M</i>) from a linear trendline.</li> </ul>
2. Calculate the variations in enthalpy.	<ul> <li>Δh<sub>n</sub> = h<sub>n</sub> - h<sub>i</sub></li> <li>Where h<sub>n</sub> is the enthalpy corresponding to a sensor reading.</li> <li>Where h<sub>i</sub> is the initial enthalpy read.</li> </ul>
3. Calculate the correction factor for every data point.	<ul> <li>F<sub>n</sub> = M × Δh<sub>n</sub></li> <li>Where F<sub>n</sub> is the correction factor for a data point.</li> </ul>
4. Calculate the new corrected VOC concentrations.	<ul> <li><i>PPB_M = PPB_sensor - F<sub>n</sub></i></li> <li>Where <i>PPB_sensor</i> is the VOC concentration reading from the sensor.</li> <li>Where <i>PPB_M</i> is the new corrected VOC concentration reading.</li> </ul>

Table 10. Overview of correction model procedure

To further clarify the procedure used in correcting a VOC reading from the AWAIR sensor, a numerical example for steps 2 through 4 is given below in Table 10 (using the same notation from Table 9). Two concentration readings from the sensor, an initial concentration reading at the beginning of the decay (say 1000 ppb) and the next concentration reading (say 900 ppb), are given. Example corresponding enthalpy values to those concentration readings are given as well (computed by the temperature and relative humidity read by the sensor). In this example, the initial enthalpy value ( $h_i$ ) would be  $32.4 \frac{Btu}{lb}$ . This value would be used to subtract from each enthalpy value ( $h_n$ ) in the data set. The initial concentration would therefore not be corrected, which would also be the case for any concentration reading which had an enthalpy value equal to the initial enthalpy value. In the case of the following concentration reading (900 ppb) of the example, it would have a corrected concentration value as there is a difference in its enthalpy value from the initial value. That difference  $(\Delta h_n = h_i - h_2)$  would then be multiplied by the calculated slope for that set of data  $\left(say M = 65 \frac{ppb}{\frac{Btu}{lb}}\right)$ . This would then be the correction factor  $(F_n)$  which would be subtracted from the original concentration reading  $(PPB\_sensor)$  to give the corrected concentration  $(PPB\_M)$ .

PPB_sensor (ppb)	$\frac{h_n}{\left(\frac{Btu}{lb}\right)}$	$\frac{\Delta h_n}{\left(\frac{Btu}{lb}\right)}$	<i>F<sub>n</sub></i> ( <i>ppb</i> )	PPB_M (ppb)	$\begin{pmatrix} M \\ \begin{pmatrix} ppb \\ \hline Btu \\ \hline lb \end{pmatrix}$
1000	32.4	32.4 - 32.4 = 0	$0 \times 64 = 0$	1000 - 0 = 1000	65
900	32.6	32.6 - 32.4 = 0.2	$0.2 \times 64 = 12.8$	1000 - 12.8 = 987.2	

*Table 11. Example of correction model procedure* 

This approach, and the result of implementing the correction model is shown in Figure 19.



Figure 19. Corrected VOC data

The newly corrected VOC plot represents the VOC concentration unaffected by enthalpy (combining the effects of temperature and humidity). Since the number of data points was reduced with the implementation of the correction model, a trendline was needed to characterize the entire period of decay for each experiment. According to existing studies tasked with developing an empirical formula for the behavior of VOC decay (discussed later in this section), the function used to describe the decay of VOCs is a double exponential decay function as seen generally with Equation (4).

$$A_1 e^{-bx} + A_2 e^{-dx} + y_o$$
; where  $b = \frac{1}{t_1}$  and  $d = \frac{1}{t_2}$  Equation (4)

A double exponential function is able to represent the combination of the initial steep decay and latter gradual decay seen in overall VOC decays. It is important to note that the  $y_o$  value is incorporated to represent the fact that there is a background level of VOCs already in the house and that the decay will not necessarily go to zero.

As stated previously, there is available literature that provides some insight into an empirical model that can describe the decay of VOCs and support the use of a double exponential decay function curve fit with a  $y_o$  value. One paper by Zhang et al. [51] was tasked with the creation of an empirical model of VOC emissions from indoor building emissions in an environment chamber. The paper derived an equation which resulted in a double exponential decay. However, it did not include a  $y_o$  because it assumed that the chamber did not have background VOCs [51]. In another paper that derived an empirical model for prediction of VOC concentration, did so considering a real indoor environment. This study did consider background VOC levels and derived a double exponential formula with a  $y_o$  value [52].

Using the same data set selected to illustrate the implementation of the correction model in this section, Figure 20 shows the result of using a double exponential decay function as a curve fit.

Also included is a separated set of data used to verify the curve fitted corrected data. The verification data used is the data points from the original sensor readings where the enthalpy is the same as the initial enthalpy value. All of the verification plots for the results of the VOC experiments are given in *Appendix E*.



Figure 20. Curve fit of corrected data including verification data

As can be seen with Figure 20, the curve fit function touches the majority of the validation data sections (PPB\_Equal\_H) while reasonably representing the decay trend following the VOC injection.

#### 4.4 Comparison of corrected data and uncorrected data

Now that the respective curve fit is found, representing the corrected VOC data, it can be compared to the original sensor VOC data as shown in Figure 21.



Figure 21. Uncorrected sensor VOC readings versus corrected data

The curve fit representing the corrected VOC data is able to show the estimated true decay of the VOCs in the test house without any of the humps caused by variations in enthalpy. This is in addition to shifting the VOC data to represent the estimated true decay of VOCs by removing the effects of enthalpy, which can be seen by comparing the curve fits of the corrected (PPB\_corrected) and uncorrected data (PPB\_sensor\_fit).

## 4.6 Conclusion of climatic correction model

Through available literature review, it was known that VOC sensor readings varied depending on variations in temperature and humidity unrelated to actual changes in VOC concentrations. This is in addition to an independent observation presented in this thesis in the form of a baseline conducted in the UTT test home where no VOCs were injected that showed variations in VOC sensor readings due to temperature and humidity. In this case, a correction model was developed to remove those effects and produce the actual trends in VOC concentrations. The two environmental variables (temperature and humidity) are combined into a single variable (enthalpy) to improve the performance of the correction model. The procedure for implementing the correction model and its development were discussed in this section. Overall,

the correction model was able to produce a new, reasonable data set of VOC concentrations with minimized variations due to environmental variables. This is in addition to a curve fit that could characterize the newly corrected set of data.

# CHAPTER 5 – METHODOLOGY AND RESULTS FOR THE EVALUATION OF PCO EFFECTIVNESS

#### 5.1 Testing facilities for VOC experiments

The VOC experiments were conducted in one of two research houses at the University of Texas at Tyler (UTT) named test house #2 (characteristics given in *3.1 Testing Facilities*). For the VOC experiments, Figure 23 illustrates the layout of the IAQ sensors used in test house #2.

## **5.2 Overview of AWAIR IAQ sensors**

The IAO sensors used in the VOC experiments include the AWAIR Element residential monitor. Three of the five available variables the sensors can read are used which includes temperature, relative humidity, and VOC concentration. Both temperature and relative humidity are read using the complementary metal-oxide semiconductor sensor (CMOS) included in the AWAIR device. The technical accuracy of the temperature readings are  $\pm 0.2$  °C with a resolution of 0.015°C, while the technical accuracy of the relative humidity readings is  $\pm 2\%$  RH with a resolution of 0.01% RH. VOCs concentrations are read using the Sensirion SGP30 multi-pixel metal oxide gas sensor integrated in the AWAIR device with a technical accuracy of 15% with a 1ppb resolution. It is important to note that VOC readings given by the AWAIR device are total concentrations of VOCs in the air as the sensor does not give the concentrations of individual VOCs. The list of VOCs the AWAIR monitor can read is an extensive but not exhaustive list that is listed on the manufacturer's website [53]. The AWAIR monitors record values every 5 minutes with data being constantly uploaded to the AWAIR cloud [54]. In terms of recalibration, before every VOC experiment, the AWAIR Element sensors underwent a reset procedure which is discussed in Appendix C.

The AWAIR sensors were placed on a stool in the center of every room excluding the kitchen and bathroom as seen in Figure 22. Due to the open concept of the test house, one AWAIR sensor was placed in the living room while another was set in the dining room area. The horizontal and vertical placement of the AWAIR sensors in each room was a result of two of reference guidelines for measuring VOCs indoors (see *Appendix B*). Figure 3 shows the location of the AWAIR sensors with respect to the house layout.



Figure 22. Stool setup for the AWAIR IAQ sensors



Figure 23. AWAIR sensor location and nearest supply registers in Test house #2

### **5.3 Experimental methods for VOC injection**

The experiments differentiate themselves in terms of the VOCs that were injected, the power status of the PCO, and the PCO that was tested. For each VOC injected, the decay of VOC concentrations in the test house was observed under the operation of the PCO air cleaner being evaluated and when no PCO device was operating. This allowed for a direct comparison to be made of the decay of VOC spikes due to the PCO air cleaners themselves versus being due to the infiltration of fresh air diluting the VOCs. Two PCO air cleaners, A and B, were evaluated for each VOC, whose specifications are detailed in *5.5 Specifications of tested PCO air cleaners*. Both PCO devices being evaluated were installed in a plenum setup, attached to the air handling unit (AHU) located in the attic of the house which is shown in Figure 24. The plenum houses the air cleaner being tested and allows for the quick exchange of different air cleaners from experiment to experiment. Each air cleaner came in its own frame which included an in-duct filter.



Figure 24. Plenum setup used to evaluate PCO air cleaners

The VOCs used included two products containing VOCs and one pure VOC substance. The manner in which each VOC was introduced or injected into the air was different. However, the overall experimental procedure remained the same. In the experiments where no PCO air cleaners were evaluated, VOCs were injected in every room in order to reach an overall concentration in the whole house. Once that concentration was reached, the VOC concentrations were left to decrease due to natural decay. Before each experiment, the PCO air cleaner being evaluated would be installed in the plenum and powered on with the HVAC fan running in order allow the catalyst to regenerate and remove VOC residue from previous experiments for a period of 2.5 hours. The PCO air cleaner being evaluated would then be removed from the plenum and placed in the other test house that did not have VOCs injected into it, so that the VOCs could be injected in test house 2. Once the VOCs injected in every room raised the overall concentration in the house to the desired value, the PCO air cleaner being evaluated was re-installed in the plenum and powered on to observe the acceleration in VOC decay due to the PCO air cleaner. Table 12 is a T-chart summarizing the general procedure of the experiments.

PCO OFF	PCO ON
• VOCs are injected in the test house	• PCO is powered on, installed in the plenum, with the AHU fan running to regenerate the PCO catalyst for 2.5 hours
• Desired concentration is reached	<ul> <li>PCO is removed so that VOCs can be injected into the test house</li> </ul>
• VOCs are allowed to decay naturally	<ul> <li>Desired concentration is reached</li> <li>PCO is re-installed in the plenum and powered on</li> </ul>

Table 12. General procedure of experiments for PCO on/off status

The first VOC product used to evaluate the PCO air cleaner was a flat black oil-based paint by Rust-Oleum with a high VOC content of 410 g/L, according to the product's safety data sheet (SDS) listed on the manufacturer's website [55]. This is in comparison to water-based or "low VOC" paint which generally has less than 50 g/L [56]. The specific list of VOCs that are included in the paint are not given in the SDS sheet as the document lists only a hydrotreated light distillate (a combination of hydrocarbons). The SDS sheet of the distillate also does not give the specific VOCs included. However, the most common VOCs used in oil-based paint solvents include toluene, xylene, ethanol, and acetone as well as many others [57]. A picture of the paint is shown in Figure 25.



Figure 25. Rust-Oleum Flat black oil-based paint [58]

The paint was applied on pieces of cardboard laid out on top of poster boards. The poster boards were placed in front of each room's AWAIR sensor. Three rounds of paint were injected in the following order in the test house: master bedroom, dining room, living room, bedroom 3, and bedroom 2. In the first two rounds, approximately 40% of the cardboard was painted over each round. In the last round, the remaining 20% was painted over. In rooms where the VOC concentration was not at the desired concentration, additional paint was applied on the cardboard pieces and surrounding poster board to reach the desired VOC concentration. The setup and progression of paint applications are shown in Figure 26.



Figure 26. Setup for paint VOC injection

The second VOC product used to evaluate the PCO air cleaner was the Mr. Clean allpurpose cleaner with summer citrus. The safety data sheet (SDS) of the product does not give the specific percentage of VOC content and instead states that, "Products comply with US state and federal regulations for VOC content in consumer products [59]." The CPID (Consumer Product Information Database) gives the complete list of ingredients in the product. Some of the ingredients are considered as chemicals of concern with some ingredients being direct VOCs themselves including Sodium hydroxide, Geraniol, Hexyl Cinnamal, Citral, Citronellol, d-Limonene, and Linalool. The product is shown in Figure 27.



Figure 27. Mr. Clean all-purpose summer citrus [60]
As seen in Figure 28, a Styrofoam board piece was laid flat on a poster board in front of each AWAIR sensor in their respective areas. The Mr. Clean all-purpose cleaner was poured and spread over the board periodically for multiple cycles in the same order as the paint injection in order to reach the desired concentration in the whole house.



Figure 28. Setup for Mr. Clean all-purpose cleaner VOC injection

The last VOC product used to evaluate the PCO air cleaners was a spray bottle of pure dlimonene at 93-95% from Lab Alley [61]. The d-limonene bottle and spray bottles used to apply the substance in the house are shown in Figure 29.



Figure 29. d-limonene bottle and spray bottle

For the d-limonene application, a white poster board piece was laid flat on the floor in front of each AWAIR in each room as seen in Figure 30. The d-limonene was first poured in the spray bottle. The spray bottle was then used to spray the VOC on the poster pieces periodically for multiple cycles in the same order as the paint and all-purpose cleaner injections in order to reach the desired concentration in the whole house.



Figure 30. Setup for pure d-limonene VOC injection

In summary, each VOC product was injected to be evaluated without the operation of a PCO and with the operation of PCO air cleaner A and B respectively for a total of nine experiments as seen in Table 13. The table also includes the duration of each experiment.

Experiment	Description	<b>Duration (hours -</b>
No.		days)
1	Paint VOC injection without PCO air cleaner operating	144 - 6.0
2	Paint VOC injection with PCO air cleaner A operating	140 - 5.8
3	Paint VOC injection with PCO air cleaner B operating	72 - 3.0
4	All-purpose VOC injection without PCO air cleaner operating	144 - 6.0
5	All-purpose VOC injection with PCO air cleaner A operating	72 - 3.0
6	All-purpose VOC injection with PCO air cleaner B operating	70 - 2.9
7	d-limonene VOC injection without PCO air cleaner operating	140 - 5.8
8	d-limonene VOC injection with PCO air cleaner A operating	120 - 5.0
9	d-limonene VOC injection with PCO air cleaner B operating	140 - 5.8

Table 13. Description of VOC experiments with duration

# 5.4 Experimental conditions of the research houses during testing

This section specifies the influential factors relevant to the experimental conditions of each test. Each room in test house 2 has a ceiling fan and at least one supply and return register. Since the testing will consider the VOC concentration of the whole home, all ceiling fans in the house were powered on in addition to the HVAC system being powered on to allow for good air mixing within the test house. All interior doors were left open in contrast to all exterior doors remaining shut throughout each experiment. In terms of the thermoregulation of the house, the HVAC was set on 99 °F in cooling mode to ensure that the compressor would not be powered on during testing. In preliminary VOC monitoring of the test house, it was seen that the powering on of the compressor would lead to unwanted spikes in VOC readings which were affected by climatic variables. Therefore, the temperature and humidity in the house was dependent on the climate outdoors. The initial temperature and humidity values were recorded for each experiment and are presented in *5.7.1 Comparison of natural decays to decays under the operation of a PCO*. It is also important to note that the 24-hour period allocated for the calibration of the AWAIR sensors (*see Appendix C*) is also regarded as the ventilation in between VOC experiments.

# 5.5 Specifications of tested PCO air cleaners

As stated in *5.3 Experimental methods for VOC injection*, the two PCO air cleaners included a frame, a filter, and the PCO core itself. The PCO cores differed in size and light intensity with PCO air cleaner A's core being bigger than PCO air cleaner B's. The specifications of the two PCO air cleaners are shown in Table 14. Both wavelengths were provided by the manufacturers of the PCO air cleaners. In the case of PCO A, the highest wavelength output of the bulb is given. In case the case of PCO B, a range in wavelength that PCO operates in was provided.

<b>Design Factor</b>	PCO A	PCO B
Size (L x H x W in inches)	16.5 x 13.5 x 1	20 x 1.875 x 1
Lamp Type	Not specified	UVA
Lamp power (Watts)	96	55
Number of lamps	528 LEDs	1 lamp
UV wavelength (nm)	395	315-350
Attached filter (MERV)	Not specified (13)	Carbon filter (16)
Catalyst	Titanium dioxide	Titanium dioxide

Table 14. Specifications of PCO A and B

The cores of the two PCO air cleaners, individually and in the plenum, are shown in Figures

31 and 32.





Figure 31. PCO A core (left) and in plenum (right)



Figure 32. PCO B core (left) and in plenum (right)

## **5.6 Evaluation of PCO air cleaner methodology**

In review of fan-driven air cleaning technologies, Zhang et al. [40] were tasked with comparing the results of different single-pass tests and chamber tests related to PCO air cleaners. They go on to state that, "the performance of air cleaners is best measured and compared with the use of Clean Air Delivery Rate (CADR)." CADR is the product of the single-pass removal efficiency and the volumetric flowrate going through the air cleaner. According to the ASHRAE journal, CADR can also be calculated using Equation (5), where *V* is the volume of the test space and  $(L_{on} - L_{off})$  is the difference between the first order loss rates  $(hr^{-1} \text{ or } min^{-1})$  for periods when the PCO air cleaner was powered on and off respectively [62]. Since each VOC experiment was represented by a double exponential function, the curve fits can be linearized by taking the natural logarithm of those curve fits. These linearized logarithmic functions can then be represented as linear functions (y = mx + b), where the slopes (m) can then be used as the corresponding first order loss rates (L).

$$CADR = V \left( L_{on} - L_{off} \right)$$
 Equation (5)

### 5.7 Results and discussion of VOC experiments

5.7.1 Comparison of natural decays to decays under the operation of a PCO

Even though variations in temperature and humidity (based on diurnal cycles of indoor VOC concentrations) were removed using the correction model, there are differences in initial values of temperature and humidity that will be a factor in the characterization of each VOC decay related to seasonal effects in VOC concentration. Thus, leading to the different characterization of each VOC decay. Higher temperatures lead to an increase in VOC emission and diffusion rates concurrently with higher levels of humidity that lead to decreased levels of adsorption affecting decay rates seen in each VOC experiments [63]. In order to consider the effects of temperature and humidity on VOC decay rates, the initial values of temperature and humidity for each VOC experiment are listed in Table 15 (provided in the order that the experiments were conducted).

VOC	PCO status	Initial temperature (°F)	Initial humidity ratio $\left(\frac{lb_w}{lb_{da}}\right)$
	PCO OFF	82	0.013
Paint	PCO ON: A	87	0.017
	PCO ON: B	86	0.012
Mr. Clean all- purpose cleaner	PCO ON: B	83	0.011
	PCO ON: A	83	0.012
	PCO OFF	82	0.013
	PCO ON: A	77	0.01
d-limonene	PCO OFF	79	0.012
	PCO ON: B	66	0.005

Table 15. Initial temperature and humidity values for all VOC experiments

In this section, Figures 33 - 35 show the direct comparison of the exponential decays representing the different VOCs used to evaluate the PCO air cleaners. The time domain of the plots was selected based on the VOC experiment that had the smallest duration to focus on their performance at the same time instances, respectively. Each decay begins at the point at which the

whole home VOC concentration was at or near the desired level. Table 16 shows the initial VOC concentration of each experiment.

VOC (Desired concentration)	PCO status	Initial VOC concentration (ppb)
	PCO OFF	4,224
Paint (4,000 ppb)	PCO ON: A	4,085
	PCO ON: B	4,125
	PCO ON: B	918
Mr. Clean all-purpose cleaner (1,000 ppb)	PCO ON: A	870
	PCO OFF	804
	PCO ON: A	1,964
d-limonene (2.000 ppb)	PCO OFF	2,118
	PCO ON: B	2,133

Table 16. Initial VOC concentrations for all VOC experiments

The paint VOC experimental results are shown and compared in Figure 33. It is worth mentioning that there are certain characteristics of the paint VOCs that will give insight into the decays seen that are explained in this section. Even though the house was ventilated for 24 hours in-between experiments, certain characteristics of the paint VOCs allowed them to accumulate in the house from experiment to experiment over time. The fact that the paint is oil-based means that it has low-water solubility which limits its adsorption onto water-droplets present indoors, increasing its ability to persist in the air [64]. Another reason for their persistence could be through the adsorption of the paint VOCs on indoor materials which can later be desorbed back into the space. This also leads into the reasoning for heightened levels of VOCs seen at the end of the decays. The baseline concentration of VOCs in the house varied in between 300 and 400 ppb, yet

PCOs. As mentioned in the *Introduction*, it is also possible that there could be some formation of VOC by-products from the PCO. However, without a direct comparison between the air immediately entering and leaving the PCO, it is hard to distinguish whether VOC by-products or the characteristics of the paint VOCs led to the heightened steady-state levels.



Figure 33. VOC decay comparison during paint VOC injection experiment

The Mr. Clean all-purpose VOC experiment results are shown and compared in Figure 34. It seems that as the experiments moved away from the paint experiments conducted first, the steady-state concentrations at the end of the decays decreased along with further ventilation of the house from experiment to experiment. The VOCs from the all-purpose cleaners were much less persistent in the air and less able to evaporate into the house to reach desired VOC concentrations (< 1,000 ppb). It seems that decays of PCO B and A were affected, respectively in order of relevance, by the VOCs leftover from the paint experiments conducted first. No major differences in initial temperature and humidity were seen in the all-purpose cleaner VOC experiments with standard deviations of ( $\pm 0.5$  °F) and ( $\pm 0.001 \frac{lb_w}{lb_{da}}$ ), respectively.



Figure 34. VOC decay comparison during Mr. Clean all-purpose VOC injection experiment

This, however, was not the case for the d-limonene VOC experiments, shown in Figure 35. The steady-state concentrations seen at the end of the decays were closer to the household baseline VOC average which seems to be due to the lower temperatures seen in the experiment. The highest drop in initial temperature and humidity was seen in the PCO ON: B period of the d-limonene test which looks to have played a factor in why the concentration reached a low steady-state. Quantitatively, the increase in the standard deviation of temperature ( $\pm$  5.7 °F) and humidity ( $\pm 0.003 \frac{lb_w}{lb_{da}}$ ) from the Mr. Clean illustrates the previous point.



Figure 35. VOC decay comparison during d-limonene VOC injection experiment

Referring back to the paint VOC experiments, keeping in mind that the accuracy of the VOC sensor readings is 15%, there is no visible acceleration of VOC abatement that is seen in the exponential plots due to the powering on of either of the two PCOs. The main differences seen are the differences in steady-state concentrations seen at the end of the decays. In the Mr. Clean allpurpose cleaner VOC experiments, neither of the decays during the powering on of the VOC were lower than the natural decay of VOCs with no PCO powered on. Rather they were significantly higher. As discussed earlier, this could be the result of remnant paint VOCs from the previous experiments oversaturating the PCO reaction surface giving a higher load on the Mr. Clean allpurpose cleaner experiments done first than the latter ones. A factor of which would also affect the periods in which the PCOs were allowed to regenerate (mentioned in 5.3 Experimental method for VOC injection) leading to the reduction of PCO capabilities. It should be noted that the influence of humidity in the Mr. Clean all-purpose cleaner experiments was minimal and would not cause the major differences in the steady-state concentrations that the curves settled at. The average temperature (81.47 °F), absolute humidity (72.73 grains/lb), and relative humidity (RH) (45.3%) seen throughout the Mr. Clean experiments had standard deviations of 0.12 °F, 2.34 grains/lb, and 1.27 %, respectively. Lastly, with regards to the d-limonene VOC experiments, no significant difference is seen between the powering on of PCO A and the PCO OFF, with the exception of the PCO ON: B in the initial decay. However, due to the initial temperature and humidity in the PCO ON: B period being noticeably lower than the others, the quicker decay cannot be solely attributed to PCO B.

Figures 36-38 show the linearized decays of the VOC experiments. As mentioned in 5.6 *Evaluation of PCO air cleaner methodology*, the decay curves (seen in the previous section) are linearized by taking the natural logarithm of each curve. The linearized curves are then

characterized by a linear function (y = mx + b), where the slopes (m) from each function are taken as the first order decay rates that are needed and used to calculate CADR in the section 5.7.2 *Evaluation of PCO performance*.



Figure 36. VOC linearized decay comparison during paint VOC injection experiment



Figure 37. VOC linearized decay comparison during Mr. Clean all-purpose VOC injection

experiment



Figure 38. VOC linearized decay comparison during d-limonene VOC injection experiment

The total slope (*m*), that characterizes the linearized double exponential decay, is the sum of both decay rates in the original double-exponential decay curves. In other words, the total slope is the summation of *b* and *d* from Equation (3) in *4.3 Methodology for VOC correction model*. The total slope will be dominated by one of the two decay rates if one is considerably larger than the other. In the case of the paint and d-limonene, as seen in Figures 36 and 38, the decay slopes for the PCO ON periods are larger than the PCO OFF period, which would indicate some level of acceleration of VOC abatement. However, this is not the case for the Mr. Clean VOC experiment. It seems that the total slope for the decays of the PCOs in the Mr. Clean VOC experiments are being dominated by the larger secondary decay rates (*b*) will be used in calculating their CADR to try to minimize the effect of the significant differences between steady-state concentrations.

# 5.7.2 Evaluation of PCO performance

In order to provide quantitative differences in between VOC experimental periods, CADR and single-pass removal efficiency are provided. This will also allow for a comparison to be made between the PCOs tested in this thesis study and others reported in literature. It must be mentioned that CADR is primarily used in closed and controlled chamber testing therefore the effects of initial temperature and humidity, and infiltrations should be considered. Table 17 gives the CADR and single-pass efficiencies for both PCOs. The single-pass efficiency is computed, as explained in *5.6 Evaluation of PCO air cleaner methodology*, by dividing the CADR with the volumetric flowrate going through the PCO (803 CFM) when the fan is powered on at 100% (according to the diagnostic portal of the HVAS system as stated in 3.3 Overview of ozone experiments).

VOC	РСО	CADR (CFM)	Single-Pass Removal Efficiency (%)
Paint	А	100	12.4
	В	2	0.3
Mr. Clean all- purpose cleaner	А	-343	-42.8
	В	-180	-22.4
d-limonene	А	124	15.4
	В	48	6

Table 17. CADR and Single-Pass Removal Efficiency of PCO A and B

The single-pass efficiency in Table 17 shows the ability or inability of the PCO device to accelerate the decay rate of the spike in VOCs injected into the test house. A positive single-pass efficiency in the table would indicate an acceleration in VOC abatement, while a negative value would indicate a deacceleration. For the paint and d-limonene VOCs, PCO A was found to have the higher single-pass removal efficiency than PCO B, considering differences in initial temperature and humidity, and residual VOCs. This distinction is representative of differences in design factors of the two PCO air cleaners, as described in *5.5 Specifications of test PCO air cleaners*. When oxidizing VOCs, PCOs with UV wavelengths closer to 315 nm would require less time for reaction due to a higher photon energy level [65]. While the PCO B would produce UV irradiance that would be more "efficient" in oxidizing VOCs, the PCO A has a bigger surface area for reactions to take place in addition to a higher number of UV lamps leading to a higher level of irradiance

increasing the rate of photovoltaic reactions. These factors, confirmed by the test results, would lead to PCO A having a design more suitable for higher removal efficiencies. A higher level of irradiance leading to a higher removal efficiency is an observation also seen in Zhong et al.'s [36] testing of in-duct PCOs (2.2.2 Laboratory testing of PCO air cleaners) where gradual increases in VOC removal efficiency were observed due to increases in irradiance intensity.

According to the literature, PCO CADR and removal efficiencies vary significantly based on the scale of the experiment, experimental design, PCOs tested, and the injected VOCs. Referring back to the systematic evaluation of in-duct PCOs performed by Zhang et al. [36], mentioned in the 2.2.2 Laboratory testing of PCO air cleaners, three difference types of PCO setups (in combination with filters) were evaluated using 8 different pure-VOCs. The positive single-pass removal efficiencies ranged from 3 to 40%. It is important to note that the PCOs evaluated in the study were exposed to a mixture of VOCs as opposed to laboratory studies with singular VOCs, which is more representative of a real indoor environment. A factor which played a role in how the different VOCs reacted on the surface of the photocatalyst.

Also mentioned in *section 2.2.2*, through Shayegan et al.'s [41] testing of PCO air cleaners in different scaled experiments, was how the removal efficiencies of VOCs decreased as experimental conditions became more representative of real-world conditions. Thus, being the reasoning for low removal efficiencies seen in this thesis study in comparison to higher removal efficiencies seen in laboratory experiments. With respect to the values seen in the Mr. Clean allpurpose cleaner experiments, it would imply that the PCOs are actually worsening indoor air conditions as opposed to improving them. Rather, as discussed throughout the paper, the VOCs leftover from the previous paint VOC experiments led to the oversaturation of the PCO photocatalysts and an increase in the steady-state levels of VOC concentration in the test house during the PCO ON periods. Thus, not painting an accurate picture of the performance of the two PCOs.

Considering the 15% uncertainty of the VOC sensor readings, differences in initial temperature and humidity, and the accumulation of certain VOCs from experiment to experiment, the single-pass removal efficiencies of the PCOs are not high enough to suggest real effectiveness in the acceleration of VOC abatement. An observation clearly seen in the decay plots of the different testing periods (PCO OFF and ON). Further reasoning on the factors related to the low removal efficiencies include the flow rate of the air passing through the PCO and the photocatalyst used. The high airflow rate seen in the test house led to significant reductions in the calculation of removal efficiency for the PCOs which stems from less residence time for the VOCs to react with the PCO. Also, apart from being oversaturated in the Mr. Clean all-purpose cleaner VOC experiments, the reactions from the PCOs and their photocatalysts were not enough in being able to show a significant acceleration in VOC abatement that could be seen in the exponential decay plots.

#### **5.8** Conclusions from VOC experiments

When comparing the exponential decays of the experimental periods, no significant acceleration of VOC abatement was seen with the powering on of the PCO air cleaners. Clean Air Delivery Rate (CADR) was used to provide a quantitative value of the PCO's effectiveness which was also be used in the computation of single-pass removal efficiency. In the paint and d-limonene paint VOC experiments, PCO A had higher removal efficiencies of 12.4 and 15.4%, respectively. The removal efficiencies of the Mr. Clean all-purpose cleaner VOC experiments were not considered due to significant differences in the steady-state concentrations seen in the test home. Different factors, including PCO irradiance and surface area of reaction, led to PCO A having

84

higher removal rates than PCO B. Keeping in mind differences in initial temperature and humidity, residual VOCs from previous experiments, and the 15% uncertainty of the sensor, the removal efficiencies from either of the PCOs are not enough to suggest high VOC abatement effectiveness in the home.

It must be noted that there are findings from the VOC experiments related to experimental design that can be improved and considered to aid the future real-world testing of PCO air cleaners. First is the spacing and order of VOCs experiments. To remove the effects of residual VOCs bleeding back into the test environment from previous experiments, it is recommended that more time be given for the ventilation of the test space in between experiments (24 hours in this study). While determining an approximate time period of ventilation using the sorption properties of the VOCs and major materials in the house, the most feasible approach would be to observe the baselines before and after VOC injections. Another finding is the suggested order in VOCs injected. Products with VOCs that are considered less persistent should be tested last. Reconsidering the order of VOCs in the VOC experiments would have led to the Mr. Clean allpurpose cleaner being used first, the pure d-limonene being used second, and the paint being used last. This is in addition to a reconsideration on the regeneration procedure of the PCO catalyst. If the VOC experiments in this thesis were to be repeated, it would be recommended that the PCO be powered on during the 24-hour ventilation period of the test home to further reactivate the PCO catalyst.

The second finding is the use of the climatic correction model to remove variations in initial temperature and humidity. This was key in defining the decay curves for all the different VOC experiments and being able to compare them to each other. Overall, the most important factor that led to the decay curves being significantly different was the order of VOCs and time of ventilation

in-between VOC experiments. A methodology to accurately compare VOC experiments that differ noticeably in initial temperature and humidity would also be beneficial in future PCO field testing. Overall, there are a lot of trends related to experimental design and results seen in the data that confirm previous experiments performed in the literature in addition to findings that address the real-world testing of PCOs.

### CHAPTER 6 - CONCLUSIONS AND RECOMMENDATIONS

The improvement of poor IAQ in residential buildings requires an assessment of indoor contaminants present indoors, in addition to the selection or combination of different mitigation strategies. Such strategies include ventilation, the use of high efficiency filters, and air cleaners or air purifiers. Assessing the effectiveness of these mitigation strategies is important for the development of effective IAQ mitigation plans for residential buildings. Two air purification technologies that have been explored for their potential in improving indoor air quality were ES air cleaners and PCO air cleaners. There is limited literature on both technologies specifically measured at the breathing zone, which is the space considered for human occupancy. Furthermore, there is currently no literature regarding the evaluation of a PCO air cleaner in a whole home environment. The examination of two ES air cleaners and two PCO air cleaners in this thesis gives insight to the effectiveness or safety of these technologies in real-world applications.

The ES air cleaners used in this thesis were tested to determine the extent that they increased indoor concentration levels of ozone. Both ES air cleaners were found to increase indoor ozone levels in the test house; however, the extent to which they were increased was less than the FDA limit of indoor ozone (21CFR801). This was also the case when reviewing studies that observed the generation of ozone from in-duct ES air cleaners. Referring back to the thesis, it was also found that even in periods where the test house was mechanically ventilated that the indoor ozone levels never exceeded the FDA limit on indoor ozone. The results from the ozone experiments in this thesis provide distinct data more relevant to occupant exposure. The studies mentioned in the literature measured ozone concentrations downstream of the ES air cleaners as opposed to in-room in the thesis. The distinction of measuring ozone concentration levels in-room prevents the overestimating of ozone generation by focusing on the breathing zone. In conclusion,

the findings from the ozone experiments provide more information regarding the safety of using ES air cleaners in residential applications.

Shifting focus, the PCO air cleaners used in the thesis were tested to determine their effectiveness in accelerating the abatement of indoor VOCs. After comparing the curves with and without PCO operation in addition to quantitative values for removal efficiency, there is no evidence to support effectiveness from the PCOs in accelerating the abatement of indoor VOCs. The highest removal efficiency seen in the literature review of PCO air cleaner testing was 72.1% for a chamber tested PCO air cleaner by Weon et al. [45]. The rest of the removal efficiencies seen in the literature review ranged from 0 to 70%. Noted, the removal efficiencies seen in the literature review were related to singular VOCs, not a collection of VOCs as seen in this thesis. The main takeaway from the VOC experiments performed in this thesis, in conjunction with the literature review of PCO testing, is that the use of PCO air cleaners in real-world residential applications are not ready for practical use. Further advancements need to be made in the development of photocatalysts that can effectively abate groups of VOCs at air flowrates representative of realistic conditions (i.e., air flowrates seen with AHU fans in residential applications). Nevertheless, additional findings from the VOC experiments presented in the thesis can assist the future realworld testing of PCO air cleaners. Including suggesting more ventilation time in-between experiments, the order of VOCs, and the development of the climatic correction model. By using the correction model, variations in temperature and humidity that caused unrelated changes in VOC sensor readings were significantly reduced, allowing for a comparison of VOC decay with and without PCO operation.

### REFERENCES

- I. Maor, "Indoor Air Quality trends," BreezoMeter, 2022. [Online]. Available: https://www.breezometer.com/pdfs/2022-Commercial-Indoor-Air-Quality-Trends.pdf. [Accessed 15 December 2022].
- [2] EPA, "Introduction to Indoor Air Quality," Environmental Protection Agency, 5
   December 2022. [Online]. Available: https://www.epa.gov/indoor-air-qualityiaq/introduction-indoor-air-quality. [Accessed December 15 2022].
- [3] R. Pitarma, G. Marques and B. R. Ferreira, "Monitoring indoor air quality for enhanced occupational health," *Journal of medical systems*, vol. 41, no. 2, 2017.
- [4] EPA, "The total exposure assessment methodology (TEAM) study: Summary and analysis," U.S. Environmental Protection Agency, 1987.
- [5] EPA, "AQI Breakpoints," Environment Protection Agency, [Online]. Available: https://aqs.epa.gov/aqsweb/documents/codetables/aqi\_breakpoints.html. [Accessed December 2022].
- [6] EPA, "What is Ozone?," Environmental Protection Agenecy, [Online]. Available: https://www.epa.gov/ozone-pollution-and-your-patients-health/what-ozone.
   [Accessed Feburary 2022].
- [7] Y. Y. Z. a. G. Z. Huang, "Contributions of indoor and outdoor sources to ozone in residential buildings in Nanjing," *International journal of environmental research and public health*, vol. 16, no. 14, p. 2587, 2019.

- [8] FDA, "CFR Code of Federal Regulations Title 21," Food and Drug Administration,
   [Online]. Available:
   https://www.accessdata.fda.gov/scripts/cdrh/cfdocs/cfcfr/cfrsearch.cfm?fr=801.415
   .. [Accessed September 2022].
- [9] EPA, "Technical Overview of Volatile Organic Compounds," Environmental Protection Agency, 2022 March 2022. [Online]. Available: https://www.epa.gov/indoor-air-quality-iaq/technical-overview-volatile-organiccompounds#3. [Accessed 20 December 2022].
- [10] WHO, Indoor air quality: organic pollutants, Geneva: World Health Organization, 1987.
- [11] MDH, "Volatile Organic Compounds in Your Home," Minnesota Department of Health, 20 October 2022. [Online]. Available: https://www.health.state.mn.us/communities/environment/air/toxins/voc.htm.
   [Accessed 20 December 2022].
- [12] EPA, "Volatile Organic Compounds' Impact on Indoor Air Quality," Environmental Protection Agency, 26 August 2022. [Online]. Available: https://www.epa.gov/indoor-air-quality-iaq/volatile-organic-compounds-impactindoor-air-

quality#:~:text=Concentrations%20of%20many%20VOCs%20are,as%20ingredien ts%20in%20household%20products.. [Accessed 20 December 2022].

[13] A. C. Heely-Hill, S. K. Grange, M. W. Ward, A. C. Lewis, N. Owen, C. Jordan, G.Hodgson and G. Adamson, "Frequency of use of household products containing

VOCs and indoor atmospheric concentrations in homes," *Environmental Science: Processes & Impacts Journal*, vol. 23, no. 5, pp. 699-713, 2021.

- [14] CARB, "Hazardous Ozone-Generating Air Purifiers," California Air Resources Board, [Online]. Available: https://ww2.arb.ca.gov/our-work/programs/aircleaners-ozone-products/hazardous-ozone-generating-airpurifiers#:~:text=Because%20ozone%20reacts%20with%20some,viruses%2C%20 bacteria%2C%20and%20odors.. [Accessed 10 March 2023].
- [15] EPA, "Residential Air cleaners," July 2018. [Online]. Available: https://www.epa.gov/sites/default/files/2018-07/documents/residential\_air\_cleaners\_-\_a\_technical\_summary\_3rd\_edition.pdf. [Accessed 20 December 2022].
- [16] A. K. Kaushik and J. S. Dhau, "Photoelectrochemical oxidation assisted air purifiers; perspective as potential tools to control indoor SARS-CoV-2 Exposure," *Applied Surface Science Advances*, vol. 9, p. 100236, 2022.
- [17] D. Aydin-Aytekin, E. Gezmis-Yavuz, E. Buyukada-Kesici, C. E. Cansoy, K. Alp and D. Y. Koseoglu-Imera, "Fabrication and characterization of multifunctional nanoclay and TiO2 embedded polyamide electrospun nanofibers and their applications at indoor air filtration," *Materials Science and Engineering: B*, vol. 279, p. 115675, 2022.
- [18] S. Weon, E. Choi, H. Kim, J. Y. Kim, H. J. Park, S. M. Kim, W. Kim and W. Choi,
   "Facet Exposed TiO2 Nanotubes Photocatalyst Filter for Volatile Organic Compounds Removal: From Material Development to Commercial Indoor Air

Cleaner Application," *Environmental Science & Technology*, vol. 52, no. 16, pp. 9330-9340, 2018.

- [19] H. Wang, X. Li, X. Zhao, C. Li, X. Song, P. Zhang and P. Huo, "A review on heterogeneous photocatalysis for environmental remediation: From semiconductors to modification strategies," *Chinese Journal of Catalysis*, vol. 43, no. 2, pp. 178-214, 2022.
- [20] EPA, "Guide to air cleaners in the home," May 2022. [Online]. Available: https://19january2017snapshot.epa.gov/sites/production/files/2014-07/documents/aircleaners.pdf. [Accessed 20 December 2022].
- [21] EPA, "Indoor Air Quality Glossary," Environmental protection agency, [Online]. Available: https://sor.epa.gov/sor\_internet/registry/termreg/searchandretrieve/glossariesandke ywordlists/search.do?details=&vocabName=IAQ%20Glossary. [Accessed 11 May 2023].
- J. Steenburgh, "Lessons in boundary layer meteorology," Wasatch Weather Weenies, [Online]. Available: https://wasatchweatherweenies.blogspot.com/2017/10/lessons-in-boundary-layermeteorology.html. [Accessed Feburary 2022].
- J. D. C. P. B. H. A. L. a. S. S. W. Jacob, "Considerations for Atmospheric Measurements with Small Unmanned Aircraft Systems," *atmosphere*, vol. 9, no. 7, p. 252, 2018.

- [24] H. T. V. A. G. B. R. B. D. C. J.-M. G. A. N. P. C. O. H. D. a. O. S. J. Petetin, "Diurnal cycle of ozone throughout the troposphere over Frankfurt as measured by Mozaic-IAGOS commercial aircraftdiurnal cycle of ozone throughout the troposphere," *Elementa: Science of the Anthropocene*, vol. 4, p. 129, 2016.
- [25] X. Z. a. S. L. Lu, "Chapter 2 Tropospheric ozone interacts with weather and climate," *Air Pollution, Climate, and Health,* pp. 14-46, 2021.
- [26] H. A. Al-Jeelani, "Diurnal and seasonal variations of surface ozone and its precursors in the atmosphere of Yanbu, Saudi Arabia," *Journal of Environmental Protection*, vol. 5, pp. 408-422, 2014.
- [27] D. K. P. a. C. Q. Lai, "Study of Outdoor Ozone Penetration into Buildings through Ventilation and Infiltration," *Building and Environment*, vol. 93, no. 2, pp. 112-118, 2015.
- [28] N. H. M. H. M. A. D. a. B. W. Ma, "Impacts of building envelope design on indoor ozone exposures and health risks in residential houses," *http://dx.doi.org/10.2139/ssrn.3972626*, 2021.
- [29] M. Y. T. K. T. a. F. H. Okubo, "Electric Air Cleaner Composed of Nonthermal Plasma Reactor and Electrostatic Precipitator," *IEEE Transactions in Industry Applications*, vol. 37, no. 5, pp. 1505-1511, 2001.
- [30] Z. S. W. L. C. C. W. J. Z. L. S. A. X. C. C. a. L. S. Lu, "Effect of granular activated carbon pore-size distribution on biological activated carbon filter performance," *Water Research*, p. 177, 2020.

- [31] K. J. a. D. J. H. Boelter, "Ozone generation by indoor, Electrostatic Air Cleaners,"
   *Aerosol Science and Technology*, vol. 27, no. 6, pp. 689-708, 1997.
- [32] A. E. L. F. L. M. J. R. S. S. J. C. W. W. P. Z. S. a. Z. J. Afshari, "Electrostatic Precipitators as an Indoor Air Cleaner—A Literature Review," *Sustainability*, vol. 12, p. 8774, 2020.
- [33] EPA, "Residential air cleaners: A technical summary," Environmental Protection Agency, [Online]. Available: https://www.epa.gov/indoor-air-quality-iaq/guide-aircleaners-home. [Accessed December 2022].
- [34] D. G. R. D. a. P. A. K. Poppendieck, "Ultrafine Particle Removal and Ozone Generation by In-Duct," *Environ. Sci. Technol*, vol. 48, pp. 2067-2074, 2014.
- [35] R. J. W. R. A. a. R. E. D. Allen, "Characterization of potential indoor sources of ozone," *American Industrial Hygiene Association Journal*, vol. 39, no. 6, pp. 466-471, 1978.
- [36] L. Zhong, F. Haghighat, C. S. Lee and N. Lakdawala, "Performance of ultraviolet photocatalytic oxidation for indoor air applications: Systematic experimental evaluation," *Journal of Hazardous Materials*, vol. 261, pp. 130-138, 2013.
- [37] W. Chen, J. S. ZHang and Z. ZHang, "Performance of air cleaners for removing multiple volatile organic compounds in indoor air," in *ASHRAE Transactions*, 2005.
- [38] J. Disdier, P. Pichat and D. Mas, "Measuring the effect of photocatalytic purifiers on indoor air hydrocarbons and carbonyl," *Journal of the Air & Waste Managment Association*, vol. 55, no. 1, pp. 88-96, 2011.

- [39] O. Debono, F. Thevenet, P. Gravejat, V. Hequet, C. Raillard, L. Lecoq and N. Locoge, "Toluene photocatalytic oxidation at ppbv levels: kinetic investigation and carbon balance determination," *Applied Catalysis B: Environmental*, vol. 106, no. 3-4, pp. 600-608, 2011.
- [40] Y. Zhang, J. Mo, Y. Li, J. Sundell, P. Wargocki, J. Zhang, J. C. Little, R. Corsi, Q. Deng, M. H. K. Leung, L. Fang, W. Chen, J. Li and Y. Sun, "Can commonly-used fan-driven air cleaning technologies improve indoor air quality? A literature review," *Atmospheric Environment*, vol. 45, no. 26, pp. 4329-4343, 2011.
- [41] Z. Shayegan, F. Haghighat and C.-S. Lee, "Photocatalytic oxidation of volatile organic compounds for indoor environment applications: Three different scaled setups," *Chemical Engineering Journal*, vol. 357, pp. 533-546, 2018.
- [42] N. Costarramone, C. Cantau, V. Desauziers, C. Pecheyran, T. Pigot and S. Lacombe,
   "Photocatalytic air purifiers for indoor air: European standard and pilot room experiments," *Environmental Science and Pollution Research*, vol. 24, pp. 12538-12546, 2017.
- [43] J. Kolarik and P. Wargocki, "Can a photocatalytic air purifier be used to improve the percieved air quality indoors?," *Indoor air: International Journal of Indoor Environment and Health*, vol. 20, no. 3, pp. 255-262, 2010.
- [44] T. M. Mata, A. A. Martins, C. S. C. Calheiros, F. Villanueva, N. P. Alonso-Cuevilla,
  M. F. Gabriel and G. V. Silva, "Indoor Air Quality: A Review of Cleaning Technologies," *Environments*, vol. 9, no. 9, p. 118, 2022.

- [45] S. Weon, E. Choi, H. Kim, J. Y. Kim, H. J. Park, S. m. Kim, W. Kim and W. Choi, "Active {001} Facet Exposed TiO2 Nanotubes Photocatalyst Filter for Volatile Organic Compounds Removal: From Material Development to Commercial Indoor Air Cleaner Application," *Environmental Science & Technology*, vol. 52, no. 16, pp. 9330-9340, 2018.
- [46] TCEQ, "CAMS 82 Monthly Summary Report," tceq.texas.gov, 30 January 2018.
   [Online]. Available: https://www.tceq.texas.gov/cgibin/compliance/monops/monthly\_summary.pl?cams=82. [Accessed 11 March 2023].
- [47] M. R. Garcia, A. Spinazze, P. T. Branco, F. Borghi, G. Villena, A. Cattaneo, A. D. Gilio, V. G. Mihucz, E. G. Alvarez, S. I. Lopes, B. Bergmans, C. Orlowski, K. Karatzas, G. Marques, J. Saffell and S. I. Sousa, "Review of low-cost sensors for indoor air quality: Features and applications," *Applied Spectroscopy Reviews*, vol. 57, no. 9-10, pp. 747-779, 2022.
- [48] A. N. Abdullah, K. Kamarudin, S. M. Mamduh, A. H. Adom and Z. H. M. Juffry, "Effect of Environmental Temperature and Humidity on Different Metal Oxide Gas Sensors at Various Gas Concentration Levels," in *IOP Conference Series: Materials Science and Engineering*, Bangkok, 2020.
- [49] Jaycon, "UNDERSTANDING A GAS SENSOR," Jaycon Systems, [Online].
   Available: https://jayconsystems.com/blog/understanding-a-gas-sensor. [Accessed 20 January 2023].

- [50] ASHRAE, ASHRAE fundamental handbook IP, American Society of Heating, Refrigerating and Air-Conditioning Engineers, 2017.
- [51] Y. Zhang, Y. Liu, Y. Wu, Y. Niu, J. Jiang and H. Guan, "Double exponential decay emperical model for indoor building materials TVOC emmissions," in *IOP conference series: Earth and Environmental Science*, Chongqing, 2019.
- [52] M. V. Ellacott and S. Reed, "Development of Robust Indoor Air Quality Models for the Estimation of Volatile Organic Compound Concentrations in Buildings," *Indoor Built Environment*, vol. 8, pp. 345-360, 1999.
- [53] AWAIR, "What Chemicals (TVOCs) Does Awair Element Detect?," getawair.com,
   [Online]. Available: https://support.getawair.com/hc/en-us/articles/360039240913 What-Chemicals-TVOCs-Does-Awair-Element-Detect #:~:text=1%2C1%2C1%2DTrichloroethane,Ethanol. [Accessed 30 December 2022].
- [54] AWAIR, "Technical Accuracy of Awair Element Sensors," getawair.com, [Online].
   Available: https://support.getawair.com/hc/en-us/articles/360060843933 Technical-Accuracy-of-Awair-Element-Sensors. [Accessed 30 December 2022].
- [55] Rust-Oleum, "Safety Data Sheet," rustoleum.com, 16 December 2022. [Online].
   Available: https://www.rustoleum.com/MSDS/ENGLISH/7776502.pdf. [Accessed 30 December 2022].
- [56] A. C. Government, "Paint: Key Environmental & Health," acgov.org, [Online].
   Available: https://www.acgov.org/sustain/documents/PaintFactSheet.pdf.
   [Accessed 30 December 2022].

- [57] Use. When worried?," P. Soto. "Paints for Indoor should I be https://www.poison.org/, [Online]. Available: https://www.poison.org/articles/paints-for-indooruse#:~:text=Oil%2Dbased%20paints%20contain%20mineral,and%20acetone%2C %20among%20many%20others. [Accessed 30 December 2022].
- [58] Rust-Oleum, "STOPS RUST® SPRAY PAINT AND RUST PREVENTION: Protective Enamel Brush-On Paint," rustoleum.com, [Online]. Available: https://www.rustoleum.com/product-catalog/consumer-brands/stopsrust/protective-enamel?ls=7776502&lc=Flat%20Black. [Accessed 2 January 2023].
- [59] CPID, "Mr. Clean, Antibacterial Cleaner, Summer Citrus-05/11/2020," whatsinproducts.com, 11 May 2020. [Online]. Available: https://www.whatsinproducts.com/brands/show\_msds/1/24056. [Accessed 2 January 2023].
- [60] ACE, "Mr. Clean Summer Citrus Scent All Purpose Cleaner Liquid 45 oz," acehardware.com, [Online]. Available: https://www.acehardware.com/departments/home-and-decor/cleaning-anddisinfectants/all-purposecleaners/1000649?store=15745&gclid=Cj0KCQiA5NSdBhDfARIsALzs2EA39Ca pdMBp6W6aLUXD6p-M8UvMP7dWl6WR0tYuh92rWFHMSHJSLdQaAlacEALw\_wcB&gclsrc=aw.ds.

[Accessed 4 January 2023].

- [61] L. Alley, "D-Limonene, Natural Orange Oil," laballey.com/, [Online]. Available: https://www.laballey.com/products/d-limonene-natural-orange-oil. [Accessed 2 January 2023].
- [62] B. Stephens, E. T. Gall, M. Heidarinejad and D. K. Farmer, "Interpreting Air Cleaner Performance Data," *ASHRAE Joural*, 2022.
- [63] S. B. Holøs, A. Yang, M. Lind, K. Thunshelle, P. Schild and M. Mysen, "VOC emission rates in newly built and renovated buildings, and the influence of ventilation a review and meta-analysis," *International Journal of Ventilation*, vol. 18, no. 3, pp. 153-166, 2018.
- [64] M. Peng, L. M. Vane and S. X. Liu, "Recent advances in VOCs removal from water by pervaporation," *Journal of Hazardous Materials*, pp. 69-90, 2003.
- [65] R. J. W. Y. W. Y. G. A. C. C. E. M. R. S. Tapia-Brito E, "Experimental study of the purification performance of a MopFan-based photocatalytic air cleaning system.," *Building and environment*, vol. 240, p. 110422, 2023.
- [66] S. Högdahl, "Placing VOC Sensors for Assessing Air Quality," KTH royal institute of technology, 2018.
- [67] EPA, "A Guide to Siting and Installing Air Sensors," Environmental Protection Agency, 15 June 2022. [Online]. Available: https://www.epa.gov/air-sensortoolbox/guide-siting-and-installing-air-sensors. [Accessed 4 January 2023].

## Appendix A – Ozone monitor zero check

To ensure the quality of the ozone monitor data, the ozone monitor was periodically zeroed out throughout the experiments. The zero check is performed by zeroing the ozone monitor around a reference point of zero ozone with the use of an ozone scrubber provided by the manufacturer (see Figure 39). The ozone scrubber is attached to the ozone monitor reading input (see Figure 40). The ozone monitor is then calibrated to read 0 ppb with the ozone scrubber on. The ozone scrubber is then removed to make readings, and the calibration is complete.



Figure 39. Ozone scrubber



Figure 40. Ozone scrubber attached to ozone monitor

### Appendix B – Justification for the horizontal and vertical placement of the ozone monitor

The location of the ozone monitor was determined as the result of a stratification test and a test that measured the critical point of ozone in the test house. The ozone monitor was placed on a stool at the center of three bedrooms and set to read background indoor ozone levels. The results are shown in Table 18.

Room	Ozone (ppb)
Master Bedroom	2.8
Bedroom 2	2.9
Bedroom 3	2.7

Table 18. Critical Point of ozone Test

The room location of the ozone monitor seemed to minimally affect the ozone readings. For this reason, the sensor was placed in a centralized location in the house in the living room.

The stratification test measured ozone concentrations at different heights from the ground. The ozone monitor was raised in increments of 11.25 inches by being placed on ladder steps. The results are shown in Table 19.

Height (Inches)	Ozone (ppb)
11.25	2.8
22.50	3.0
33.75	3.5
45.00	3.6
56.25	3.6
67.50	3.4

Table 19. Stratification of ozone test

The critical point of ozone was around 45-56 inches from the ground. This is considered "waistline" level. For this reason, the ozone monitor was placed on a stool in that height range.

#### Appendix C – AWAIR sensor recalibration process

Technical information on the AWAIR IAQ sensors were obtained from online documents and direct conversation with the manufacturer. Based on the methodology of VOC testing in the study, the manufacturer recommended that the AWAIR sensors be manually reset every week or for every different test with different chemicals. In order to reset the multiple AWAIR sensors, they must be exposed to fresh air in the same room in order to have the same reference baseline. The AWAIR IAQ sensors were placed in the living room of the test house with open windows in addition to mechanical ventilation being powered on (see Figure 41). Exposed to fresh air, the AWAIR IAQ sensors read their lowest possible value of VOC concentration which is close to 20 ppb. At that point, the AWAIR Element sensors are reset through the AWAIR mobile phone application where the option is found. While the official calibration of the sensor took 2-3 hours, it was recommended by the manufacturer to keep the sensors exposed to fresh air for a total of 24 hours. Once the sensors were reset, a notification was given indicating that the sensors were calibrated. They were then placed in their designated rooms with the windows closed (after the initial 3-hour period) and the mechanical ventilation being powered off (after a 24-hour period).



Figure 41. Resetting of the AWAIR IAQ sensors in the living room/dining room area

## Appendix D – Justification for horizontal and vertical placement of AWAIR IAQ sensors

A study was conducted on the placement of VOC sensors for assessing Indoor Air Quality (IAQ). In the study, a simple model was developed to conclude how well different areas of a room correspond to the average VOC concentration of the occupied zone, given that occupants are the main source of VOC. Through literature review and the application of their model in two locations, an apartment and an office, some guidelines for the placement of sensors were developed. Some relevant guidelines from the study include [66]:

- 1. Sensors may be placed outside of the occupied zone, from a VOC perspective
- 2. VOC sensors should not be placed in corners, as particles tend to collect due to the noslip principle and more stagnant air.
- Particles collect in the ceiling, given the geometry and ventilation of the case studies, and the average concentration might be considerably higher there compared to other height averages
- 4. The best placement in theory is in the middle of the room, where none of the occupants touch or breath on the sensor, while VOC metering at walls may lead to spikes due to non-uniform behavior in particle concentration.
- 5. Measuring IAQ a few centimeters out from the wall or ceiling could result in values significantly closer to the average of the occupied zone

The EPA has similar guidelines for indoor IAQ sensor placement with the main suggestions being to install sensors out in the open, allow free air flow around sensor (at least 180 degrees), keep away from windows and doors, and to place near the typical breathing height (3 to 6ft) [67]. The guidelines and recommended actions listed above lead to the AWAIR Sensors being placed on chairs close to the 3 to 6 feet height range in centralized locations in their respective rooms.

# Appendix E – Verification plots for climatic correction model

Figures 42-50 show the results of decay fitting of the VOC injections experiments explained in *Correction model for climatic variables affecting VOC readings* presented in *Methodology and results for the evaluation of PCO effectiveness*.



Figure 42. Paint VOC injection – PCO OFF



Figure 43. Paint VOC injection – PCO ON – PCO A



**Appendix E – Verification plots for climatic correction model (Continued)** 

Figure 44. Paint VOC injection – PCO ON – PCO B



Figure 45. Mr. Clean all-purpose cleaner VOC injection – PCO OFF


**Appendix E** – Verification plots for climatic correction model (Continued)

Figure 46. Mr. Clean all-purpose cleaner VOC injection – PCO ON – PCO A



Figure 47. Mr. Clean all-purpose cleaner VOC injection – PCO ON – PCO B



**Appendix E – Verification plots for climatic correction model (Continued)** 

Figure 48. d-limonene VOC injection – PCO OFF



Figure 49. d-limonene VOC injection – PCO ON – PCO A



**Appendix E** – **Verification plots for climatic correction model (Continued)** 

Figure 50. d-limonene VOC injection – PCO ON – PCO B

## **Bio-sketch**

Giovanni Cerrato was born in Palestine, Texas and has lived in east Texas for 23 years. He is a recent graduate from the University of Texas (UT) at Tyler. He graduated with a bachelor's degree in mechanical engineering and is seeking his master's degree in mechanical engineering at UT Tyler in the academic school year of 2022-2023. His main area of research is in the space of indoor air quality topics and technologies. His aim is to work in the HVAC industry.